Edith Cowan University Research Online

Theses: Doctorates and Masters

Theses

1-1-2000

Regular teachers' attitudes to the need for additional classroom support for integrated students with intellectual disabilities

Rory McNally Edith Cowan University

Follow this and additional works at: https://ro.ecu.edu.au/theses

Part of the Special Education and Teaching Commons

Recommended Citation

McNally, R. (2000). *Regular teachers' attitudes to the need for additional classroom support for integrated students with intellectual disabilities*. Edith Cowan University. Retrieved from https://ro.ecu.edu.au/theses/1355

This Thesis is posted at Research Online. https://ro.ecu.edu.au/theses/1355

Edith Cowan University

Copyright Warning

You may print or download ONE copy of this document for the purpose of your own research or study.

The University does not authorize you to copy, communicate or otherwise make available electronically to any other person any copyright material contained on this site.

You are reminded of the following:

- Copyright owners are entitled to take legal action against persons who infringe their copyright.
- A reproduction of material that is protected by copyright may be a copyright infringement. Where the reproduction of such material is done without attribution of authorship, with false attribution of authorship or the authorship is treated in a derogatory manner, this may be a breach of the author's moral rights contained in Part IX of the Copyright Act 1968 (Cth).
- Courts have the power to impose a wide range of civil and criminal sanctions for infringement of copyright, infringement of moral rights and other offences under the Copyright Act 1968 (Cth).
 Higher penalties may apply, and higher damages may be awarded, for offences and infringements involving the conversion of material into digital or electronic form.

Regular teachers' attitudes to the need for additional classroom support for integrated students with intellectual disabilities

by

Rory Mc Nally, B.A., B.Ed., H.Dip. Ed.

A 4-Unit thesis submitted in partial fulfillment of the requirements for the award of Master of Education

Supervisors: Professor Peter Cole

Dr. Russell Waugh

May 2000

EDITH COWAN UNIVERSITY LIBRARY

USE OF THESIS

The Use of Thesis statement is not included in this version of the thesis.

Declaration

I certify that this thesis does not, to the best of my knowledge and belief:

- incorporate without acknowledgement any material previously submitted for a degree or diploma in any institution of higher education;
- contain any material previously published or written by another person, except where due reference is made in the text; or
- contain any defamatory material.

| Signed: | |
|---------|--|
| | |
| | |

Date: 000

Acknowledgements

I would like to thank Professor Peter Cole for his guidance, enthusiasm, patience and humour. The greens await you! Thank you to Dr. Russell Waugh for stepping into the breech and fine-tuning the thesis. I gratefully acknowledge the help of Dr. Amanda Blackmore in the fledgling stages of this study. I would like to thank my family and my supportive wife, Jo, for granting me the time and the environment to complete this thesis. Thank you to David, Darragh and Declan for those vital distractions. I must make special mention of my daughter, Aoife, for her technical wizardry. Thanks also to Covey and Harry for clearing my head!

Abstract

The purpose of this study was to examine regular classroom teachers' attitudes to additional classroom support, both physical and personnel, for students with mild and severe intellectual disabilities who may be integrated into regular classrooms. The concerns of the regular classroom teachers with regard to the successful inclusion of students with intellectual disabilities are certainly merit addressing. As the regular classroom teachers are largely responsible for the education of these integrated students, it is worthwhile to examine their perceived support levels in relation to integrated students with intellectual disabilities. It is anticipated that the results of this study will prove useful when decisions are being made by educational administrators about the level and type of support needed for integrated students with intellectual disabilities. The results should be beneficial as a guideline for those concerned with the appropriate allocation of funding for students with intellectual disabilities who are educated in regular classrooms. As the subjects of this study were Catholic teachers employed in Catholic primary schools in Perth, Western Australia, it is anticipated that the results of this study will prove useful to the administrators in the Catholic Education Office of Western Australia.

The dependent variables were physical and personnel support. The independent variables were ability, effort and school. Seventy-two classroom teachers from six East –Metropolitan Catholic Primary Schools in

Perth, Western Australia, were used as participants. These teachers were presented with a vignette describing a hypothetical male student. Ability (average, mild, severe) and effort (low, moderate, high) were systematically varied to create a nine-cell design. Vignettes were randomly assigned to the 72 teachers. The teachers were asked to respond to two measurements for the dependent variables. The first comprised a seven-point Likert scale measuring their perceived need for additional physical support and additional personnel support in the regular classroom. The second was a magnitude-scaling instrument, which also required responses on additional physical and personnel support. A multivariate analysis was completed for the seven-point scale and magnitude scaling of the dependent variables. Wilk's criterion indicated no significant multivariate interaction between the factors of ability and effort. The multivariate analysis yielded a main effect for ability for both dependent variables.

Univariate analyses showed that teachers perceived a significant difference between students with and without intellectual disabilities in the levels of additional personnel support needed but no significant difference between students with mild and severe intellectual disabilities in the levels of additional physical support. The ability level of the students was the critical variable that determined the levels of additional support, as perceived necessary by the regular classroom teacher. Regular classroom teachers did not perceive the factor, effort, as being significant. Consistent with attributional research findings, they did not perceive student effort as needing additional classroom support. Teachers perceive a strong need for

personnel classroom support being necessary for students with mild and severe intellectual disabilities.

There is a paucity of global research specific to the expected levels of additional physical and personnel classroom support needed for students with intellectual disabilities and none in Australia. This study has implications for the allocation of resources within schools in that it may offer guidelines for determining the levels and type of support given to regular classroom teachers so that integrated students with intellectual disabilities may succeed in the regular class.

| | Table of Contents | PAGE |
|------------------|--|------|
| TITLE PAGE | | 1 |
| USE OF THESIS | | 2 |
| DECLARATION | | 3 |
| ACKNOWLEDGEM | ENTS | 4 |
| ABSTRACT | | 5 |
| TABLE OF CONTE | NTS | 8 |
| LIST OF TABLES | | 10 |
| LIST OF FIGURES | | 11 |
| LIST OF APPENDIC | CES | 12 |
| CHAPTER 1 | INTRODUCTION | 13 |
| | Background | 13 |
| | Inclusion | 13 |
| | Support | 16 |
| | Ability | 18 |
| | Focus of this study | 19 |
| | Research questions | 20 |
| | Structure of the thesis | 20 |
| CHAPTER 2 | LITERATURE REVIEW | 22 |
| | Global perceptions and models of inclusion | 23 |
| | Attribution theory | 32 |
| | Theories of teachers' attitudes to support | 35 |
| | Conclusion | 53 |

| CHAPTER 3 | METHODOLOGY | | 55 |
|----------------|---|---|----------------------|
| | Measurement and v selection | ariable | 55 |
| | Measurement tools | | 56 |
| | Trial and resulting a | mendments | 61 |
| | Subjects | | 62 |
| | Design | | 63 |
| | Procedure | | 65 |
| | Response rate | | 66 |
| | Hypotheses | | 67 |
| CHAPTER 4 | RESULTS | | 68 |
| | Outliers | | 68 |
| | Reliability and Vali | dity | 70 |
| | Multivariate results variables for both n | for dependent neasurement tools | 71 |
| | Seven-point scale – comparison tests | univariate and | 72 |
| | Magnitude scaling - comparison tests | - univariate and | 77 |
| | Summary of results | | 81 |
| CHAPTER 5 | DISCUSSION | | 82 |
| | Summary of study | | 82 |
| | Conclusions Limitations of studi | ies | 83 84 |
| | Implications for: | Administrators Teachers Schools Research | 85 87 88 88 |
| LIST OF REFERE | ENCES | | 90 |

•

List of Tables

| TABLES | | Page |
|--------|---|------|
| 4.1 | Multivariate test of main effects and interactions for schools/ability/effort | 71 |
| 4.2 | Multivariate test of main effects and interactions using the seven-point scale for ability/effort | 72 |
| 4.3 | Multivariate test of main effects and interactions using magnitude scaling for ability/effort | 77 |
| 4.4 | Comparison of ability levels for the seven-point data | 75 |
| 4.5 | Comparison of ability levels for magnitude scaling | 78 |

List of Figures

| FIGURES | | Page |
|---------|--|------|
| 3.1 | Study Design | 64 |
| 4.1A | Score on the physical items for ability/effort levels for the seven-point scale | 73 |
| 4.1B | Score on the personnel items for ability/effort levels for the seven-point scale | 74 |
| 4.2A | Score on the physical items for ability/effort levels for magnitude scaling | 80 |
| 4.2B | Score on the personnel items for ability/effort levels for magnitude scaling | 80 |

List of Appendices

| | | Page |
|---|---|------|
| Α | Study explanation | 96 |
| В | Consent form | 97 |
| С | Instructions to teachers | 98 |
| D | Vignette | 100 |
| E | Seven-point measurement on physical support | 101 |
| F | Seven-point measurement on personnel support | 103 |
| G | Magnitude scaling measurement on physical and personnel support | 104 |

Chapter 1

INTRODUCTION

Background

There has long been an awareness among educators that students with intellectual disabilities can be successfully educated in mainstream classes. This push for the inclusion of these children into regular schools, which began some 40 years ago, has brought with it many concerns for regular classroom teachers. How do regular classroom teachers view the whole inclusion movement? It must be difficult for regular classroom teachers who disagree with the concept of inclusion to successfully include students with intellectual disabilities into their class. What additional support is necessary and available to the regular classroom teacher so that integrated students with intellectual disabilities can be successfully educated? Do regular classroom teachers see a difference between levels of ability? Is the level of effort expended by the student an important factor in determining the expectations of the classroom teacher in regard to student performance?

Inclusion

The term inclusion means that a student with disabilities may be partially or fully included in the regular classroom, dependent upon the severity and number of disabilities and the level of additional support available for that student (Fuchs & Fuchs, 1994). In this study, inclusion does not mean full inclusion, where the inclusive school educates all

students in the mainstream, as advocated by Stainback and Stainback (1992). Proponents of this type of inclusion claim that the placement of students with disabilities, irrespective of the number or severity of the disability, in any educational setting other than the regular classroom, puts these children at risk of an inferior education and deprives them of social relationships. In this study, inclusion does not mean "…the *full* inclusion of *all* persons with disabilities in *all* aspects of societal life" (Lipsky & Gartner, 1991. p.52. italics in original).

How much time does a student with disabilities have to spend in the regular classroom in order for inclusion to take place? There has been a great deal of debate on this topic and Brown, Schwarz, Udvari-Solner, Kampschooer, Jolenson, Jorgenson and Greenwald, (1991) state that the student with disabilities need not spend all of his or her time in the regular class, but that that regular class should be viewed by all teachers and students as the class to which this student belongs. They also claim that " It is better to be an 'insider' who goes out for short periods of time, than it is to be an 'outsider' who comes in," (p. 40).

Evans (1994) claims that it is not enough to decree that all students with disabilities be placed in the regular classroom. If inclusion is to work, there must be a wide variety of personnel to support the regular classroom teacher to provide a healthy educational environment for all students.

The inclusion movement started in the early 'sixties' under the name 'mainstreaming', changing its name to 'integration' in the late 'seventies'. The current trends of partial and full inclusion affect and reflect regular teachers' attitudes to inclusion (Fuchs & Fuchs, 1994; Stainback &

Stainback, 1992). The rights of students with disabilities in the United States and England to as normal an education as possible are enshrined in Law (Public Law 94-142 and the 1981 Education Act respectively). In Australia there is a growing awareness of the needs and acknowledgements of the rights of people with disabilities (Center & Ward, 1987). Principles of normalisation, the inclusion of students with disabilities into the regular classroom and the practice of individualized education programmes are now commonplace within the Australian educational system. Normalisation is the creation of as normal as possible a learning and social environment for the student with disabilities (Kirk, Gallagher & Anastasiow, 1997). While many educators would like to entrench such policies in a legal mantle, successive governments have managed to side step the issue, preferring to pay lip service to such philosophies.

With the impact of behavioural technology in the U.S and the immense progress towards the principles of normalization in Scandinavia (which emphasized the need for support structures in education and within the local community), it was no longer possible to exclude the severely handicapped from being educated. In 1971, the Senate Standing Committee in Australia recommended that the educational needs of disabled students and the training of teachers to meet these needs be shouldered by the Commonwealth Government. In 1972, the Interim Committee for the Australian Schools Commission was established and it recommended that grants be made available to train teachers to meet the needs of the handicapped and that finance be given to one university in each state to establish diagnostic and research facilities in the area of special education.

Research in Australia suggests that regular classroom teachers generally hold less favourable attitudes towards inclusion than do administrators and other professional staff (Center & Ward, 1987). However, the gap between teachers' attitudes and those in administrative positions is closing (Harvey, 1992). Inclusion has long been a controversial topic here and, little wonder, as we are so often swayed by the winds of change blown by American and European academia. A recent study in America, involving over 7000 regular classroom teachers, found that 65% of regular teachers supported the concept of inclusion, 54% were willing to include students with disabilities in their own regular class and 55% of regular teachers felt that inclusion provided benefits to the student with disabilities (Scruggs & Mastropieri, 1996). While the educational rights of students with disabilities have been legislated for in some other countries (such as UK and USA), it seems that the concerns of the personnel most responsible for the practice of such rights (classroom teachers) have yet to be addressed.

Support

Regular classroom teachers are educational personnel who are appropriately qualified and experienced, are currently employed as schoolteachers by the school authorities to teach regular school grades in regular classrooms. The regular classroom teacher has considerable responsibility for the success of a student with an intellectual disability enrolled in a regular class (Garvar-Pinhas & Schmelkin, 1989) and has many concerns about the inclusion of students with intellectual disabilities

into a regular class. These teachers often claim that they need additional support to help students with intellectual disabilities succeed in the regular classroom. These additional supports are over and above the supports that are normally available to the regular classroom teacher. Additional_supports may comprise a time allocation for the regular teacher to plan for the inclusion of the special needs student, professional development of the regular teacher, personnel support, material resources, administrative support, class size and consideration of the severity of the disability (Scruggs & Mastropieri, 1996). A task force set up in 1994 to report on the education of students with disabilities and specific learning difficulties recommended an increase in resources for the provision of adequate services to meet the needs of students with disabilities (Education Department of Western Australia, 1994, Recommendation 61).

This study focuses on two types of additional classroom support for the regular teacher who has a student with an intellectual disability included in the regular class. Additional physical support refers any additional curriculum resources (textbooks, remediation and extension materials, enrichment programmes, computers and software), used to support specific students in the regular class. Additional personnel support refers to any additional personnel (specialist teachers, aides, paraprofessionals, volunteers,) who are employed to support students with disabilities included in the regular class.

Education institutions should be aware of the levels of support that they can offer so that appropriate placement of a student with disabilities can be matched. Center and Ward (1987) suggest a graduated model

whereby appropriate assessment of the child's school needs can be matched with the support levels offered by the school and the correct placement made.

Ability

An intellectual disability refers to a student's inability to learn because of substantial limitations in cognitive functioning. It is characterised by below average intellectual functioning and limitations in at least two of the following adaptive skills areas: communication, self-care, health, basic academics, leisure, employment, safety, home living and social skills. Intellectual disability was, up until 1992, classed as mild if the person's IQ was between 69 and 50 and severe if the person's IQ was between 40 and 20.

Kirk, Gallagher and Anastasiow (1997) point out that definitions of intellectual disabilities are not 'cast in concrete'. Factors such as cultural differences, the effect of community environments, the individual's relative strengths in particular domains and the improvement that can result from various supports must be taken into account. They define the three distinguishable levels of intellectual disability, mild, moderate and severe. Mild refers to mental development at between one-half and three-quarters of the normal rate; moderate, at one-half of the normal rate and severe as mental development as less than one-quarter of normal cognitive growth.

An understanding of these levels of ability would be very beneficial to the regular classroom teachers so that they can evaluate their expectations

of students with an intellectual disability and make more informed decisions on the levels of support necessary for these students to be fully integrated.

Focus of this study

This study investigated the perceived additional support levels for integrated students with intellectual difficulties, as identified by 72 regular classroom teachers, employed in six Catholic primary schools in the eastern metropolitan area of Perth. The independent variables comprised schools, ability and effort. The dependent variables were additional physical and personnel support. In this study, the ability variable was divided into three levels. These levels were classed as average intellectual ability, mild intellectual disability and severe intellectual disability. The effort variable was also divided into three levels; low, moderate and high. Effort refers to the level of input that a student expends in the areas of classwork, assignments and homework.

The selection of ability and effort in this study is based on attribution theory. Attribution theory focuses on when and how causal inferences are made. Heider (1958) paved the way for research into this domain believing that motivation, ability and situational factors combine to promote a certain action. Weiner (1979) extended Heider's work by identifying mood and effort as major factors that promote certain actions. The present study investigates whether effort and ability are major factors when regular classroom teachers are determining support levels for integrated students with an intellectual disability.

Research Questions

With regard to expected additional support levels (curriculum resources and personnel), do regular classroom teachers' attributions of students' effort differ for students with and without intellectual disabilities? Is effort a crucial factor when regular teachers make judgements on the level of support necessary for particular students?

Is there a perceived difference for teachers in the expected levels of additional support for students with mild and severe intellectual disabilities? With regard to ability, do regular classroom teachers see a greater need for additional support for students with severe disabilities as opposed to students with moderate disabilities?

Is there a difference in additional physical and personnel support among the levels of ability? It was considered that there would be no difference among the schools and therefore no research questions were developed on this factor.

Structure of the thesis

This four-unit thesis is divided into five chapters. Chapter one provides an introduction to the study. It briefly describes factors that may influence regular classroom teachers' perceptions of the inclusion of students with intellectual disabilities into the regular classroom.

Chapter two presents a literature review and looks at the research on global perceptions of inclusion in Europe, America, Asia and Australia,

focusing on concerns for adequate support for the inclusion of students with intellectual disabilities. It gives a succinct history of attributional theory and its relevance to this study, before examining the research on regular teachers' attitudes to additional support for integrated students with intellectual disabilities.

Chapter three describes the methodology used in the study. It describes the measurement tools used, which comprised responses to a vignette by way of a seven-point scaled questionnaire and magnitude scaling. The chapter then describes the subjects in the study, the procedure by which the data were collected and the nine-cell design of the study. Chapter three ends by describing the hypotheses developed for the study.

Chapter four describes the results of the study. It explains why the data were examined for outliers. It describes how the data were analysed using multivariate procedures and why the schools factor was not included in subsequent univariate analyses. Tables and charts depict these results, which are interpreted and explained.

Chapter five summarises this study, discusses the implications of this study for research, educational administrators, classroom teachers and schools. The results are compared to the findings of similar research.

Chapter 2

LITERATURE REVIEW

This chapter is divided into five sections – search procedure, global perceptions and models of inclusion, theories of regular teachers' attitudes to additional support for students with intellectual disabilities, attribution theory and the conclusion. In the global perceptions of inclusion, the current status of inclusion in a number of selected countries across four continents will be examined and commented upon. Attribution theory is then examined in relation to this study. This will be followed by a detailed analysis of the research on regular classroom teachers' attitudes towards additional support for integrated students with intellectual disabilities. A summary and comments form the conclusion.

As this present study was conducted in Catholic schools only, it is necessary to highlight the way in which children with disabilities are treated in the Catholic system as opposed to Government schools. As in the public system, selected Catholic schools have an on-site special education unit. These special education units cater for the educational needs of students with mild to moderate intellectual disabilities. There are no Catholic schools that cater for the needs of students with severe intellectual disabilities. The students who receive support in Catholic special education units are initially enrolled in their appropriate regular class. The special education unit is run by an appropriately qualified teacher. Unlike the Government system of special education centres within regular primary schools, there are no separate administrative personnel to look after the running of the unit. The school principal assumes this responsibility. The students are funded by

Commonwealth grants and topped up by the school's budget. Special education policies in both Catholic and Government schools will be similar by 2004 as both bodies embrace the Curriculum Framework (Curriculum Council of Western Australia, 1998).

Global perceptions and models of inclusion

To demonstrate current global trends in mainstreaming students with disabilities, the researcher will provide a succinct précis of movements in Europe, Asia, America and Australia. Mainstreaming (and more recently inclusion) is identified as the integration of students with disabilities into regular education classes in order to accommodate the requirements of 'least restrictive environment' as legislated by the Education for All Handicapped Children Act of 1975 in the USA.

Europe has responded in a very positive fashion to the notion of equality of education for all students with disabilities as outlined in the American Education for All Handicapped Children Act of 1975 (Public Law 94-142). While a lot of European countries don't have the rights of students with disabilities enshrined in law, the majority of countries acknowledge these rights in various education policies. The following data on Europe were compiled by the European Agency for Development in Special Needs Education and made available on the worldwide web. In the U.K., under the terms of the Education Act of 1996, school authorities have a duty to place children with special educational needs in mainstream schools with their peers wherever possible. School authorities must ensure that the child receives the appropriate level of support in order for successful integration

to take place. All students are entitled to have their educational needs identified and to receive the National Curriculum (Education Reform Act 1988). The Code of Practice 1994 states that pupils should be involved in decision-making about their learning, including target setting, *support levels*, evaluation and reviews. These legislative changes are the direct result of the 1970 Education (Handicapped Children) Act and the Warnock Report (1978). Along with the American 1975 Public Law 94-142, these laws paved the way and defined the direction of schooling for students with disabilities in the mainstream school system.

In Portugal, the Comprehensive Law of 1986, Decree 319 of 1991 and the recent legislation of July 1997, recognise the right of equal opportunities for pupils with special needs to integration into local mainstream schools. As a result these students have the right to the adaptation of their educational environment as well as to the educational process as well. These adaptations may focus on changes to or provision of the physical environment, materials, special equipment and technical aids. The Portaria 1102/97 established the means for mainstream participation and support, Section 105/97 reinforcing the conditions for integration allocating support teachers to mainstream schools. There is a very clear tendency to support inclusion and to continue to improve conditions as confirmed by decreasing numbers attending special institutions and a steady increase in the numbers of students with disabilities attending mainstream schools.

The absence of comprehensive special education legislation in Ireland has resulted in a number of alterations to the Constitution being

developed to cater for the rights of the student with disabilities. The Report of Special Education Review (1993) recommend that: "appropriate education for all children with special educational needs should be provided in ordinary schools, except where individual circumstances make this impracticable." The Review Committee proposed a continuum of education provision to meet a continuum of special educational needs. Interestingly they propose a similar model to that used in some Australian states, the Irish version dealing with a wider spectrum of educational institutions. The recent drafting of the Educational needs. Current Government policy is to encourage the maximum possible level of inclusion of children with special needs into mainstream schools and to put into place the necessary supports to facilitate this development. It is envisaged that this support would take the form of remedial, resource and itinerant teachers.

The UN Convention on the Rights of the Child (1989) is now ratified by 177 countries worldwide. Among other concerns, Article 23 states the right of children with disabilities to special care and education opportunities, designed in a manner conducive to the child achieving 'the fullest possible social integration and individual development.' The UNESCO Salamanca Statement (1994) is a report from the UN's education agency calling on the international community to endorse the approach of inclusive schools by implementing practical and strategic changes. In June 1994, representatives of 92 governments and 25 international organisations formed the World Conference on Special Needs Education. They developed a dynamic new Statement on the education of all disabled children, which

called for inclusion to be the norm. They adopted a new Framework for Action, the underlying principle being that ordinary schools should accommodate all children, regardless of their physical, intellectual, social, emotional, linguistic or other disabilities. The Statement continues:

'Regular schools with this inclusive orientation are the most effective means of combating discriminatory attitudes, creating welcoming communities, building an inclusive society and achieving education for all.' (UNESCO Salamanca Statement. 1994)

It asks the UN to improve their networking for more efficient support to integrated special needs children and calls on countries to establish inclusive-not special-schools. The Centre for Studies on Inclusive Education (1994) issued a report on special education needs in the Asia region. The report, based on 15 country case studies, argues that children with disabilities can be integrated into the regular educational system more successfully and a cheaper cost than being placed in a segregated setting. It goes on to state that Asian countries are becoming aware of the value of inclusion, both to the pupil and the community. A lot of countries have begun to address improvements regarding the quality of education, which the participation of children with disabilities requires.

One of the poorest countries in the world, Nepal, has set a goal to integrate students with mild to severe impairments in mainstream primary education. The target is to make special education an integral part of basic primary education. Since 1985, the Nepal Association for the Welfare of the Blind has set up 21 schools, 20 of which are attached to regular schools.

In India, the Five Year Plan (1991-96) has increased the budget for children with impairments by more than five times. India supports a major national development programme on the integration of such children into regular schools.

The Philippine government has defined the ultimate goal of special education to be the integration of learners with special needs into the regular school system and eventually into the community (Section 5,Article 1 of Policies and Guidelines for Special Education).

Sri Lanka, despite its troubled past, was an early pioneer of mainstreaming. It regards the integration of children with and without impairment as one of the most important contributions to community living. Regular teachers in Sri Lanka receive a lot of in-class support from volunteer parents. Korea, Malaysia, Sri Lanka, China, Nepal, Indonesia and Thailand are among the Asian countries to have introduced Individual Learning Programmes in classrooms, which include special needs children. Special classes affiliated to regular schools enrolled 2,651 children in 1990 In China. In Nanjing province in 1993, in-service teacher training began to provide support for children with mild learning difficulties.

Such definitive changes in legislation, educational policy and practice with regard to the integration of children with disabilities have swept across Europe and Asia in the past decade. Legislation has prompted schools in the U.S. to reassess their funding procedures for students with disabilities. With the legal sword of this law hanging over their heads, much more funds have been made available to support students with disabilities in their least restrictive environment.

In 1994, the U.S. Department of Education identified 4.3 million students as having a specific learning disability. Many of these students are placed in regular classrooms for most of their instructional day. It is generally agreed that empirical evidence supports the notion of inclusion as identified from the early sixties by researchers such as Johnson (1962), Kern and Pfaffle (1962), Bacher (1964) and Diggs (1964).

School authorities in all Australian states have enunciated policies that propose that children with disabilities should be integrated wherever possible into regular schools and classes (de Lemos, 1994). The A.C.T. Schools Authority conducts two special schools for primary and junior secretary students who have moderate to severed intellectual handicaps or a developmental delay. Both schools conduct a variety of programmes affording pupils the opportunity to interact with non-disabled peers. This growing awareness of the needs for support for regular classroom teachers who have children with disabilities in their class is evidenced by the various Government funded reports into this area. The notion of equality among sections of the community was communicated in the report *A Fair Chance For All* (Department of Employment, Education and Training, 1990), the overall objective being "to ensure that Australians from all groups in society have the opportunity to participate successfully in higher education".

The Department of Employment, Education and Training (1990) study involved 18 institutions of higher education, selected because of their commitment to cater for the needs of the disadvantaged and people with disabilities. The Department of Employment, Education and Training (DEET) together with the Higher Education Council of the Board developed

a National Equity Higher Education Plan, identifying objectives for people with disabilities, which was to increase their participation in higher education, with a 12 month target of doubling the number of people with disabilities enrolled in third level institutions.

The Department of Employment, Education and Training (1990) study sought to identify good practice strategies for each disadvantaged area from the selected institutions. From these data, a composite model was to be developed and an appropriate funding model set up to underpin policy in relation to Equity funding for the educationally disadvantaged. The study confirmed that while Australia has been lauded for recognizing the rights and needs of people with disabilities, these rights are not enshrined in Australian law. The needs of such students appear so obvious while the resources dedicated to them are so small. This lack of resourcing and the absence of legal recognition of the rights of the disabled, places the status of services for students with disabilities under threat of any cost cutting exercise.

de Lemos (1994) has compiled the most comprehensive report on the status of students with disabilities in Australian schools. This report was based on a study of educational provision for students with disabilities carried out at the behest of the Australian Education Council and funded by the Commonwealth Department of Employment, Education and Training. The overall objective was to develop an understanding of the optimum way of providing school educational services to students with disabilities.

The study included a survey of schools to obtain data on what provisions are made for students with disabilities. It also included a

questionnaire to government and non-government educational authorities to determine policies and procedures in this area. I will examine these two areas in a bid to identify the level of support available to students with disabilities in the regular school and to compare the policy and practice in this area as this directly affects the perceptions of regular teachers' attitudes towards support.

Practices in special education at school level were identified from data collected by means of questionnaires to school principals, teachers and parents. The teacher responses numbered 680 at primary level and 419 at secondary level from a total of 369 schools (190 Government, 116 Catholic and 63 Independent schools). The sample was drawn from the various levels in all states. There was an over-sampling of Catholic and Independent schools in the target sample and, coupled with a very high response rate from this sector, this resulted in an over-representation in the achieved sample. This sample was then appropriately weighted.

The distribution of the sample representing the type of school attended was equivalent to the distribution of the national population of students with disabilities. Enrolment in primary schools was 45% in the population, 46% in the sample; for secondary schools 25% in the population, 26% in the sample; in special schools 27% in the population, 28% in the sample.

In primary schools, 32% of the sample group admitted that they did not have any special facilities for special needs students. 60% indicated that they had at least one of three types of special facility (ramps/modified toilet/specialist learning area), while only 8% had more than three special

facilities for such students. 60% also acknowledged that the building structure of the school caused mobility problems for students with disabilities.

Among the 117 sample primary schools, the average enrolment of students with disabilities in each school was 10. In the 107 secondary schools sampled, an average of 16 students with disabilities was enrolled in each school. Secondary schools had an extra 308 teacher and teacher aide staff in full-time employment terms for 1558 students with disabilities. In the primary schools sampled, only 45% of the additional staffing level were teachers, compared to 60% being teachers in the secondary school sample. Ten percent of the additional staffing in secondary schools comprised nonteachers specialists.

There was a big difference among school types in regard to the level of service they accessed from visiting specialists. Twenty five percent of the sample secondary schools had no special facilities for students with disabilities. Sixty six percent indicated that they had at least one of the previously mentioned types of facility, while a mere 7% listed more than three special facilities. Thirty six percent of the principals surveyed thought that their school's physical resources were adequate so as to successfully enroll and cater for the needs of students with disabilities (43% in primary schools). Again a high proportion (66%) of secondary schools have buildings not conducive to the ease of mobility for students with disabilities.

The population of Australian students who have an identified learning disability represents 2% of the student population (Ward, Center, Outhred & Pieterse, 1987). Of these 62,000 students, (accurate figures are

difficult to obtain due to inter-state differences on the definition of disability and the levels of pre-requisites necessary to access services), 27% were enrolled in special schools, 24% in special classes/units attached to primary and secondary schools and 49% were enrolled in regular primary and secondary schools. A constant challenge is to provide support for these students, especially the 49% in regular classes, who can so easily flounder. It becomes essential for educational bodies to adopt a model that ensures delivery of this support (Ward, Center, Outhred & Pieterse, 1987).

Attribution Theory

Heider (1958) examined the influence of attributions on feelings and behaviour and proposed that there was a link between people's expectations of others and their behaviour towards them. The selection and matching of ability and effort, the main independent variables used in this study, were based on attributional theory, in particular, the work of Weiner (1979) and Clarke (1997). Weiner (1979) put forward the theory that a person's motivation is determined by how well the person performed the same task in the past. The stability of this attribution, according to Weiner, determines one's expectations of future performance: "If one attains success...and...the conditions of causes of that outcome are perceived as remaining unchanged, then success...will be anticipated with a great degree of certainty" (Weiner, 1979, p. 9). Weiner considers that effort is considered internal but can be controlled. Ability is also internal but cannot be controlled. It is this stability (or lack thereof) of causes that determine the expectancy of future success (or failure). Individuals who attribute their performance to unstable

controllable causes, such as effort, tend to persist, in the belief that if they try harder, they will improve their performance. Conversely, if performance is attributed to a stable, uncontrollable cause, such as ability, they see little chance of changing their performance level in the future.

Clarke (1997) tested the validity of these attributional principles. She surveyed 97 classroom teachers. Each teacher was presented with a vignette of a hypothetical boy, indicating his level of ability and effort and whether he was intellectually disabled or not. There were four dependent measures. They were evaluative feedback, rating of anger, pity and expectations in regard to the boy. A 2 x 2 x 2 analysis of variance with repeated measures was conducted on each dependent variable. Results showed that the greatest rewards, the most pity and the least expectations were allocated to intellectually disabled students. This suggests that teachers see an intellectual disability as the single, uncontrollable cause of academic failure. Self-attributions carry over to the attributions of others (such as teachers) on individual performances (target students).

When an individual is seen as being in control of an outcome, the individual is viewed as responsible for performance. An outcome outside the control of the individual is seen as one for which the individual is not responsible (Weiner, 1986). When failure is ascribed to low ability, it is seen as resulting from a fixed characteristic, whereas failure due to lack of effort is under the individual's volitional control. Teachers' perceptions of the causal properties of their students' academic outcomes result in a number of emotional behaviours towards the students.

Weiner and Kula (1970) reported that students with low ability, who made no effort in class, received less punishment than students of high ability who made no effort. This supports the view that teachers perceive low ability as being the cause of failure as it is internal, stable and uncontrollable. Effort, while internal, is unstable and deemed to be controllable. Bar-Tal (1979) extended the Weiner model and applied it specifically to the classroom situation. The teachers' causal perceptions of student performance determine their behaviour towards the students. Like Weiner, Bar-Tal suggests that teachers base their expectations of student performance on the degree of stability and controllability of the causes. If success or failure of the student is a result of stable causes, a similar performance may be expected in the future. If performance is attributed to unstable but controllable causes, teachers might assume that a more successful performance can be achieved in the future. However, if performance is attributed to unstable and uncontrollable causes, teachers cannot predict future student performance. Research has supported the proposal that teachers' behaviour towards students is determined by their expectations regarding future student performance (Dusek, 1975; Braun, 1976 and Cooper, 1979).

Graham and Weiner (1986) confirmed a link between anger and pity and rewards and punishment. They found that the classroom teacher might feel anger towards a child who failed because of lack of effort, particularly if the child has high ability, yet feel pity for a child who has failed because of low ability. For the same low performance one child may be punished
less than the other because of the teacher attributing the cause of the failure to ability.

Graham (1990) researched teachers' perceptions of the principal causes of failure in the classroom and reported ability and effort as the two critical variables influencing the outcomes of student achievement. Effort and ability of students were established as being linked to a regular teacher's perception of their level of support (Graham, 1990). A similar design was employed in the present study.

Theories of regular classroom teachers' attitudes to additional support for students with intellectual disabilities.

Regular classroom teachers are most responsible for the success of the inclusion of students with disabilities into the regular classroom (Garvar-Pinhas & Schmelkin, 1985). This being the case, it is increasingly important to pay close attention to the concerns that general classroom teachers raise in relation to problems that they perceive as a result of having students with disabilities in their classes. However, a study carried out by Harasymiw and Horne (1976) found that although the classroom teacher bears the brunt of the responsibility for the success of the inclusion of students with disabilities, administrators are more likely to be asked about the factors identified as being critical for the success of such inclusion programmes. A considerable amount of research has been done on the attitudes of regular teachers to integration and Horne (1985) provided an extensive review of this literature, finding weighty evidence of negative attitudes to mainstreaming.

The body of research in this area highlights concerns such as handicap type, expertise, time and support levels as being concerns of regular teachers in regard to mainstreaming. General classroom teachers want to know what the most effective means of support are in order for children with disabilities to succeed in the regular classroom. They are concerned with the type, quality and regularity of support that best suits the needs of such students. Support for students with disabilities in the regular classroom can be of a physical or personal nature. Physical resources would include additional instructional material and resources, additional computer software and accessories and any other additional learning materials. Personnel support would comprise additional support staff (aides, paraprofessionals, volunteers). Adequate availability of support for students with disabilities goes some way towards assisting the general classroom teacher to effect beneficial instruction.

Larrivee and Cook (1979) reported a negative reaction towards inclusion yet here is growing evidence of a more positive attitude towards inclusion taking place among regular classroom teachers (Harvey, 1992). It would seem that the cycle is recurring as more recent studies seem to indicate a shift in support of regular teachers towards the concept of mainstreaming (Fulk & Hirth, 1994; Vaughn, Schumm, Jallad, Slucher & Saumell, 1996). It is impossible to review regular teachers' attitudes towards expected support levels for integrated students without being aware of these constant shifts and conflicts in a bid to evaluate the efficiency of inclusion. However concerns about the practicalities of inclusion still remain

only partially addressed, especially integrating children with more severe disabilities.

Teacher expertise, time constraints, support levels, type and level of disability are the major concerns of regular teachers in regard to inclusion (Center &Ward, 1987; Childs, 1981). General classroom teachers want to know what the most effective means of support are in order for children with disabilities to succeed in the regular classroom. They are concerned with the type, quality and regularity of support that best suits the needs of such students. Support for students with disabilities in the regular classroom can be of a physical or personnel nature. Physical resources would include additional instructional material and resources, additional computer software and accessories and any other additional learning materials. Personnel support would comprise additional support for students with disabilities goes some way towards assisting the general classroom teacher to effect beneficial instruction (Center & Ward. 1987).

Much research has been done on the attitudes of classroom teachers to inclusion. However, there is a dearth of studies specifically examining levels of support needed to maintain the child with disabilities in a regular classroom (Coates, 1989, Larivee & Cook, 1979). The bulk of research on inclusion has been carried out through teacher surveys in an effort to identify the concerns teachers have regarding the inclusion of the child with disabilities in the regular classroom (Gans, 1987).

A research synthesis carried out by Scruggs and Mastropieri (1996) examined 28 leading studies of teachers' perceptions of

mainstreaming/integration .In this study, Scruggs and Mastropieri (1996) searched all relevant databases for articles describing teacher attitudes towards mainstreaming or inclusion. Additional information was gleaned from relevant books, literature reviews and reports. All major special education journals were also examined. Data were deemed relevant only if they dealt with teachers' attitudes towards teaching students with disabilities in the regular classroom and if the could be presented in a format in which "percent agree" to specific relevant items could be ascertained. Altogether, 28 reports published from 1958 to 1995 were identified. Respondents included 10,560 teachers from rural, urban, suburban or combined school districts in the Northeast, Southeast, Midwest and Western parts of the United States, New South Wales and Montreal. Mean years of teaching experience was12.7 years. All surveys dealt with, among others, the issues of resources. For each item or cluster of items, an outcome of percent of respondents in agreement was derived.

A consistent finding of this study was that regular classroom teachers need support in teaching classes that include students with disabilities. These supports deemed necessary for the success of the students with disabilities included, time, training, personnel support, material resources, class size and consideration of the severity of disability. In ten of the surveys, conducted in nine states in the Northeast, Midwest, South and West of the United States, respondents were asked whether they had enough expertise and training to help a student with disabilities succeed in the regular classroom. Of the 2,900 respondents, 29.2% agreed that regular

classroom teachers could support a student with disabilities in their class as a result of adequate expertise and training.

In six of the surveys, questions related to the issue of adequacy of resources for integrated students with disabilities. These studies were conducted in the Midwest of the United States and New South Wales, Australia. There were 3,268 respondents in total. Many of these surveys distinguished between material and personnel resources. Of the respondents, 22.29% felt that they had adequate support (material and personnel) in order to support students with disabilities in the regular classroom.

In three investigations, 11% agreed that they had adequate personnel support for integrated students with disabilities, while 37.6% in two investigations agreed that they had adequate material support for integrated students with disabilities. Overall, teachers did not agree that sufficient resources were available for the successful integration of students with disabilities into the regular classroom.

Hudson, Graham and Warner (1979) surveyed 518 elementary teachers in Kansas about their attitude to inclusion. Thirty eight percent of the teachers felt that the materials they used for children with disabilities were inadequate. They noted that these additional materials were out of date, not instructionally useful and were often incomplete. Fifty eight percent reported that they felt they did not have the necessary support services available to them. By support services, teachers were referring to in-class support (aides, paraprofessionals, volunteers) and external consultative support (psychologists, therapist, resource teachers).

Larrivee and Cook (1979) devised an attitude scale to investigate the effect of classroom support variables on the attitudes of 941 regular classroom teachers toward the integration of special needs children. One of those variables was cited as "The availability of additional support services for accommodating special needs students, such as resource room, resource teacher, remedial reading teacher, counselling and appropriate instruction material (appendix, p322). The final teacher sample closely approximated the actual school population breakdown in New England. A Likert scale, reduced to 30 items after an item analysis, boasted a reliability rating of .92 (Spearman-Brown). Twenty five hundred questionnaires were mailed to 250 principals of the 250 randomly selected schools, providing a 54.4% return rate. Of the 7 variables surveyed – grade level, class size, school size, school type, success rate with special ed. students, level of administrative support and availability of additional support services – only three (success with special education. students, level of administrative support and availability of support services) had a significant impact on teachers' attitudes.

Success with special education students correlated highest with a positive teacher attitude to inclusion and with the level of support services available to the teacher (Larivee & Cooke, 1979). Findings would seem to indicate that teachers are willing to accept special needs students into their regular class if they can rely on the necessary support from other personnel and from adequate additional resources.

Childs (1981) surveyed 450 teachers from primary to high school who had integrated students with mild intellectual disabilities in their

regular classrooms. This study produced a negative attitude towards inclusion, as 50% of the teachers were unable to deliver quality instruction due to the absence of resource material and consultant services to teach these students.

Schultz (1982) used open-ended survey questions to find out the concerns of regular classroom teachers to educating children with disabilities. The responses were then categorised under the following headings: Time constraints, class size, training, resources and type of handicap. Of the 378 regular classroom teachers who responded, 99.5% cited the availability of additional instructional materials as being important.

Gallagher (1985) surveyed 466 regular classroom teachers in midwest Missouri to elicit information about their perceptions of inclusion. Only 119 teachers (25.5%) confirmed that they were aware of available resources to facilitate mainstreaming efforts. Available resources were defined as appropriate instructional materials over and above that which is normally available to the regular classroom teacher. This would seem to indicate that the availability of additional resources was not communicated to the regular classroom teacher and that the classroom teacher had not asked the appropriate personnel about such materials resources. It raises concerns about the level of planning that went into supporting the students with disabilities in regular classrooms.

Gans (1987) mailed a questionnaire to regular classroom teachers requesting information on 99 demographic and attitudinal variables related to the integration of students with disabilities into the regular classroom. Gans based the content of the questions in his questionnaire on results from

a pilot questionnaire, interviews with school personnel and a review of the literature. He drew responses from 128 regular educators and 133 special educators in 21 school districts in Ohio and was able to investigate the importance of both types of variables and compare regular and special education teachers' responses. The attitudinal variables comprised four factors. The first factor dealt with teacher confidence in setting goals, measuring achievement, behaviour management, etc. The second factor surveyed attitudes to the effects (positive and negative) of integrated students with disabilities on the classroom. The third factor dealt with whether teachers had enough time for instructional planning and the fourth factor was concerned with the level of support (material and personnel) readily available.

Twenty-one (80%) of the public school districts in three North Eastern Ohio counties agreed to participate in the Gans study. Regular educators were randomly selected once they fulfilled the criteria needed to balance gender, grade level and subject areas taught. These characteristics were thought to be influential in the formation of attitudes toward handicapintegration surveys (Larrivee & Cook, 1979). All special educators were sampled in each district into four factors (as identified above) by a principal components factor analysis. The make-up of these factors was similar for both the special and regular educators.

Results of the Gans study indicated that regular classroom teachers were influenced more by individual characteristics (especially the number of handicaps the student had) than by attitudinal variables. The reverse effect was observed for the special educators. Both groups yielded a 96%

predictive accuracy. While regular teachers scored a high 93 standardized co-efficient for the importance of the variety of handicapping conditions, they recorded a 0.0 for Factor 4, which dealt with support material available. It would appear that regular teachers find the number of handicaps that a student has much more important than being concerned with materials available. Gans also reports that the study is weakened somewhat by the fact that teachers confirmed that they were torn between how they should respond for professional reasons and their actual reasons.

Coates (1989) surveyed 94 regular classroom teachers in Iowa, in a bid to determine to what degree regular classroom teachers supported the underlying beliefs of the Regular Education Initiative. Proponents of this initiative believe that it is more beneficial for intervention to occur in the regular classroom than for a student to be taken out of the classroom. The Regular Education Initiative came about in 1985 as a result of a speech made by Madeleine C. Will, assistant secretary for the Office of Special Education and Rehabilitation Services, where she argued that 'pull-out' programmes for students with disabilities stigmatized students and resulted in lower expectations. In 1986, when Wang, Reynolds and Walberg presented a paper criticizing the pull-out approach and advocated collaborative models (eg. teacher assistance teams), the stage was set for the birth of a new initiative – the Regular Education Initiative.

Coates (1989) designed a15 item survey to measure the regular teacher's agreement or disagreement with views advocated by the Regular Education Initiative. An example of the former would be following item (and the item pertinent to this paper): "Given additional support, I would be

able to meet the educational needs of mildly handicapped students in my class without the need for a resource room". An example of an item that disagrees with the views of Regular Education Initiative is "Resource rooms are not an effective model for meeting the educational needs of mildly handicapped students" (Coates, 1989, Appendix 2a).

In the Coates study (1989), teachers were asked to respond to each item on a 5-point Likert Scale. Two open-ended queries were also included. Selected teachers were sent a survey with an explanatory letter. Subjects were given 4 weeks to return the completed survey, being contacted by phone if they had failed to do so after 5 weeks. The response rate was 75% (94 teachers).

The general trend of responses indicated disagreement with the underlying assumptions of Regular Education Initiative. Items were given a numerical value to indicate the level of agreement (1) and disagreement (5) with Regular Education Initiative philosophy. The item on support scored 3.74, indicating that these general classroom teachers, even with support, did not perceive themselves as being fully equipped to support students with mild disabilities in a regular setting. Regular teachers were particularly concerned with students who "fall through the cracks" and these students were perceived to need more one-to-one assistance.

Myles and Simpson (1989) sought information on the classroom modifications regular classroom teachers would request if they were to accept a mildly handicapped student into their regular classroom. This study was prompted by the recognition that the regular and special education systems, although associated, are largely independent of each

other. Proposals to marry both settings (Reynolds, Wang & Walberg 1987) have added fuel to the debate and as mentioned earlier, spawned the birth of the Regular Education Initiative. Myles and Simpson (1989) set out to determine the factors that would influence regular classroom teachers to actively nurture the underlying assumptions of the inclusion of students with mild disabilities into their regular classroom. They devised an instrument to elicit this information. It comprised (a) a cover letter, (b) instructions (c) a vignette of a hypothetical boy (labelled and unlabeled profiles of educable mentally handicapped boys, behaviour disordered boys, and learning disabled boys), (d) fifteen questions on mainstreaming options and (e) eight questions of a demographic nature. The vignettes were field tested by independent special education doctoral students to confirm their lucidity and accuracy.

One hundred regular education teachers took part in the Myles and Simpson (1979) investigation. They varied in age, experience, grade level taught, area of certification and district size. Twenty six percent were recruited from a Kansas mid western suburban school district. The remaining 74% were enrolled in university remedial reading and special education introductory courses. The response rate was 90% for teachers and 100% university enrollees.

Myles and Simpson randomly assigned vignettes of students labelled educable mentally handicapped, behaviour disordered and learning disabled to regular classroom teachers. They were told that this student was to be enrolled in their class and to make the minimal classroom modifications necessary that would persuade them to be confident in supporting this child

fully in the regular classroom. The list of classroom modifications was a result of a 1997 survey conducted by the National Education Association ("Teachers Opinion Poll, 1975) and from current educational trends. These included, " (a) decreased class size, (b) additional planning time, (c) assistance of a paraprofessional, (d) availability of support service, (e) consultation with a special educator, and (f) inservice workshops (p482).

In the Myles and Simpson (1979) study, half of the resulting modifications dealt with support facilities perceived necessary by the regular educators in order to support students with disabilities in the regular classroom. The teachers were also asked to compare their preferred classroom conditions to their realistic classroom situation relative to each of the above variables. Finally the teacher was asked whether they would integrate the student with or without the modifications they had suggested. This probe sought to (a) compare differences in the type and number of changes needed to convince regular classroom teachers to integrate labelled and unlabeled handicapped children into their classroom, (b) compare actual and preferred support services, (c) determine the willingness of regular classroom contingent upon identified education modifications. Seventy eight percent of the respondents selected support services and modified class size, as opposed to 27% choosing inservice training.

The factorial analysis of variance procedure yielded no significant difference between the levels of modifications. There were no differences between classification and labelling, and between actual and preferred class sizes, number of consultations services, amount of planning time,

availability of paraprofessionals and the number of in-sessions training sessions.

The surprising result indicated no significant differences between actual and preferred number of support services. When the types of support services were broken down, regular teachers indicated that special educators and psychologists were more valuable than counsellors. Interestingly, the role of counsellors and crisis teachers were least preferred of the support services. Perhaps the most significant finding is that, contingent upon consideration of their concerns regular classroom teachers are very willing to include students with disabilities.

In this Myles and Simpson (1979) study, there were no statistically significant differences between the number of modifications selected by teachers as a function of the category of student. Teachers did indicate that class size and support services were the most desirable changes if a student with a disability were to be enrolled in their class. In addition, most teachers identified the use of a paraprofessional for at least five half days per week was needed for successful integration. Teachers specified behaviour management and instructional techniques as their preferred type of consultative support service. Respondents seemed to suggest that with the proper levels of support services and resources they would assume instructional responsibility for handicapped students mainstreamed into their classroom.

Although there was no significant difference between actual and preferred support services, the fact that 78% of the teachers selected the support category as a modification suggests that they are not receiving this

support. It can also be interpreted that support personnel are not providing the type of service expected by regular classroom teachers.

Center and Ward (1987) conducted an extensive study on regular teachers' attitudes to inclusion. Of the 2,219 teachers surveyed, 74.9% supported mainstreaming for students with mild mobility or sensory disabilities. In the same survey, having sufficient resources for mainstreaming was identified as a key-determining factor influencing attitudes to mainstreaming. The respondents were asked how many of their opinions on integration would change if they had access to a greater number or frequency of support services. More than half of the group indicated that this factor alone would make them alter their attitude to integration in a more positive fashion.

The survey also sought to elicit data about teacher's attitudes towards current support services. The results indicated that the school counsellor service is considered to be the most effective service of all those provided by school in order to effect successful mainstreaming of disabled children. This is in dispute with the findings referred to earlier by American researchers, Myles and Simpson (1989).

The mean satisfaction level for all support categories for all categories of children with disabilities was extremely low (15%). The main reason for this dissatisfaction stems from the unavailability or paucity of such services (7.8% satisfaction rate in Catholic School). Because the counselling service is the only one that operates regularly, it has become overloaded, resulting in school counsellors becoming less effective and

drawing sharp criticism from teachers. It is this lack of adequate support that is associated with fostering negative attitudes towards mainstreaming.

An analysis of quantitative responses in this Centre and Ward (1987) study indicates that regular teachers require classroom aides and a decrease in class size whenever a child with a disability is included in the class. Seventy five percent of teachers approved of itinerant teachers of the intellectually handicapped, possibly indicating that they feel resource teachers on their own are unable to provide adequate support for such children. Fifty percent of the respondents were attracted by the notion of parental assistance in the classroom to support integrated children. This was particularly noticeable among Catholic teachers.

Center and Ward (1987) summarized the findings of their comprehensive study by stating that there was general dissatisfaction with the current support services offered in schools for children with disabilities. This deficiency was noted in both the quality and quantity of support services. They also point out that teachers do not necessarily want to replicate these services but are seeking alternative methods of support for the full inclusion of students with disabilities.

Parmenter and Nash (1987) documented a study on the inclusion programmes in two special schools. One school had 33 of its 66 students involved in integration, the other had 14 of its 26. Integration ranged from 3 hours per week to full time regular school attendance. Questionnaires were distributed to the staff of the two special schools (100% response rate), parents of the children in the two schools (55% response rate) and structured

interviews were conducted with 15 teachers in the regular schools in which the students were integrated.

Special education teachers noted that there is very little 'back up' assistance provided to the integrated children and they suggested that the regular class teacher also needs specialist assistance within the classroom. They also commented that integration was being conducted on the cheap as inadequate resources were being provided, especially at the regular school level. The small sample of 15 teachers from the regular schools limits the predictive validity and generalizeability of such data. Reports on the amount and type of support varied. The majority of teachers expressed dissatisfaction with the regularity of support.

The authors of this small study identified three factors important for the inclusion process. The first had to do with adequate planning, the second with support levels and the third with educating the community to accept a greater 'deviancy' among children. The authors claim that adequate resources to support the programmes of integration must be provided. These range from basic physical access to additional curriculum materials and modification of the School Curriculum. By providing access to the regular school curriculum, it is essential that the regular school receive the support of resource personnel who are adequately trained in the area of a special education.

Sigafoos and Elkins (1994), from the Schonell Research Centre in Queensland, investigated the concerns of teachers towards the integration of children with physical disabilities and multiple disabilities. The study was conducted at a regular secondary school servicing 1300 students aged 12-18

years. Seventeen teachers were selected by the principal to participate in the study. The authors acknowledge that this may have resulted in a biased and unrepresentative sample but indicated that the principal was in the best position to select teachers who had most contact with students with disabilities.

In the Sigafoos and Elkins study (1994), two questionnaires were developed, based on the Larrivee and Cook model (1979). Predictably, attitudes depended on the type and degree of disability. One pertinent outcome of the study was that the success of integrating children with multiple disabilities depends on the extent to which teachers received adequate. The emerging themes and concerns identified from analysis of teacher comments were, in order of priority, individual needs, degree of disability, disruption to classroom, peer interaction, need for support, time constraints and stress safety.

Sigafoos and Elkins (1994) note that school principals and guidance officers might lend support to the classroom teacher by way of assisting in the modification of curriculum and instruction techniques and by creating more favourable administrative arrangements. This would help make the inclusion programme be viewed in a much more positive fashion.

The Education Department of Western Australia (1993) has recently issued a new policy and guidelines for the education of children with disabilities. It specified individual policies for students with intellectual and physical disabilities, exceptional needs, autism, vision impairment and hearing impairment.

The policy states that:

"Schools will ensure that all students, irrespective of the degree of sensory, physical or intellectual disability, have the opportunity to be educated in the most educationally enhancing environment, consistent with the provision of a quality education which best meets the need of the individual student." (Education Department of Western Australia, 1993).

Education authorities in Australia have developed guidelines and policies in regard to the placement of students with disabilities into educational settings. The policies of Government Education Authorities, and Catholic and Independent Schools are all required to adhere to the dictates of recent legislation in every state concerning equal opportunity, social justice, community and health services (Education Department of Western Australia, 1993). These policies have been developed to focus on inclusion whenever possible and feasible, on parental involvement and on the need for support structures.

Australia has made major changes in regard to the educational provision of students with disabilities. These changes include the shift from the provision of these educational services in segregated settings to integrated settings and the development of the idea of inclusive schooling. There is also recognition that special educational settings may best meet the needs of some students with disabilities (de Lemos, 1994.).

de Lemos (1994) demonstrated that primary and secondary schools are more dependent on visiting staff than special schools. The majority of teachers expressed a need for more professional development activities to assist them in their task of facilitating the successful integration of students

with disabilities. Variations in curriculum focus were noted in the different settings. Major areas of concern for both principals and teachers alike had to do with staffing and being able to access support specialist services. More one-to-one contact for students with disabilities was recommended. While the philosophy of integration was generally welcomed, some schools were experiencing problems putting the policies and theories into practice. These difficulties were in the main related to staffing and resources and the lack of training for teachers who had students with disabilities in their mainstream class.

Conclusion

The body of research worldwide confirms an increase in an acceptance of the doctrine of normalisation and inclusion of students with disabilities into regular classrooms. While there are many concerns being voiced by regular classroom teachers and despite the recent global legislation in all developed countries, the problems of adequate support for the integrated students still remain. While many of the articles in this literature review cite support as being essential to integrated students with intellectual disabilities, there seems to be nothing definitive about the type or the level of support deemed necessary to support the integrated student across levels of ability. Even the comprehensive research synthesis by Scruggs and Mastropieri, while acknowledging the need for resources and support for students with disabilities, does not enter into discussion on this topic. The levels of support, both physical and personnel, are still foremost in the list of variables that affect teachers' attitudes towards the integrated student. It

would seem that this problem, identified during the birth of integration almost 40 years ago, is one that can only be solved by powers greater than the mere classroom teacher.

Yet research into regular teachers' attributions of perceived support levels for students with disabilities included in regular classes has been limited, especially in Western Australia. This paucity of research in the area of support for regular classroom teachers of students with disabilities and the absence of research specifically in the area of additional support for regular classroom teachers of integrated students with disabilities makes the present study worthwhile. It is anticipated that the results of the present study will prove useful to principals and educational administrators in Western Australia when making decisions on support levels and funding for integrated students with disabilities. Will the results confirm the findings of Weiner (1970) and Clarke (1997) in regard to ability being the defining factor that is responsible for the success of a learning disabled student in the regular classroom? Will teachers see a difference between the need for additional physical and additional personnel support? Do regular classroom teachers perceive a difference between the needs of the mild and severely intellectually disabled in regard to additional physical and personnel support? Does effort affect regular teachers' perceptions of the need for additional support for students with intellectual disabilities?

Chapter 3

Methodology

In this chapter, the selection and samples of the measurement tools are presented, the design of the study explained and the procedural methodology described. The chapter ends with the formulation and presentation of the hypotheses.

Measurement and variable selection

As the researcher was attempting to measure teachers' attitudes, this required a quantitative measurement tool. One to one and group interviews were discarded as possible measurement tools because of the time needed to conduct them, their timetabling implications and their susceptibility to researcher's subjectivity. The questionnaire was chosen because of its user friendliness to subject and researcher, the brevity of time taken to complete it (ten items in total) and its ability to be used successfully in group situations. The researcher used a seven-point Likert scale to ensure responses that could be measured more accurately. The magnitude scaling was used to confirm (or dispute) the responses of the questionnaire.

The two dependent variables selected for this study were additional physical and additional personnel support for the regular classroom teacher to assist integrated students with intellectual disabilities. These variables were selected because they have not been the subject of any such study in Western Australia, despite being cited as a major concern of regular classroom teachers who teach integrated students with intellectual disabilities (Scruggs & Mastropieri, 1996; Center & Ward, 1987).

Additional physical support refers any additional curriculum resources (textbooks, remediation and extension materials, enrichment programmes, computers and software), used to support specific students in the regular class. Additional personnel support refers to any additional personnel (specialist teachers, aides, paraprofessionals, volunteers,) who are employed to support students with disabilities included in the regular class.

Measurement tools

For the purpose of this study, the researcher developed a vignette describing a hypothetical male student. Empirical evidence indicates that the majority of students with disabilities are male (Hallahan & Kauffman, 1980). The hypothetical student, Brian, was described in two key areas (See Excerpt 3.1, p 56). It included a statement of the level of his learning ability in terms of the presence or absence of an intellectual learning disability. Where Brian was described as having an intellectual learning disability, this disability was described as being mild or severe. A statement on the typical pattern of effort expended by Brian in the classroom formed the second key area. Effort was reported in terms of a good deal of, a modest amount of or little effort.

Vignette

"Brian is a potential student for your class next term. Recent psychological testing indicates Brian has no / a mild / a severe/ intellectual disability compared to students of his age. He currently undertakes instruction in the core subject areas of Mathematics, English, Social Studies and Science. Brian participates in social activities and is aware of school rules.

He always / sometimes / rarely/ works hard in class, making a good deal of / a modest amount of / little/ effort to complete assignments and homework. He can participate in group work, likes soccer and has two pet rabbits of which he is very fond. Brian's parents are anxious that he adjusts well to his new school and hope that he can settle smoothly into his new environment."

The researcher developed a seven-point Likert scale questionnaire to measure regular teachers' responses to their perceived additional support needs for the student described in the vignette. The researcher chose a seven-point scale to give teachers more opportunity to accurately indicate the level of their response.

The questionnaire was divided into two sections, each section containing six items. Two items were subsequently deleted from the second section (See 'Trial'). The first part of the questionnaire (See Measurement A sample) asked teachers to indicate the level of additional physical support they would need in order for Brian to succeed in various regular classroom subject areas. The second part of the questionnaire (See Measurement B sample) asked teachers to indicate the level of additional personnel support they would need in order for Brian to succeed in small and large groups inside and outside the regular classroom.

<u>Measurement A sample</u> (Appendix E)

Additional physical support

1. How many <u>additional</u> resources will you need to give adequate support to Brian in **Mathematics** lessons?

| 1 | 2 | 3 | 4 | 5 | 6 | 7 |
|----------|---|---|---|---|---|-----------|
| Very few | | | | | | very many |

Measurement B sample (Appendix F)

Additional personnel support

1. How much <u>additional</u> personnel support would you need to help Brian in **outdoor activities?**

| 1 | 2 | 3 | 4 | 5 | 6 | 7 |
|-------------|---|---|---|---|---|-----------|
| Very little | | | | | | very much |

The researcher developed a magnitude scaling measurement (See Measurement C) in regard to perceived physical and personnel support levels in order to confirm the results of the questionnaire.

<u>Measurement C</u> (Appendix G)

Magnitude Scaling

- 1. Rate the additional support necessary for a newly enrolled student with average learning ability who puts in an average amount of effort.
- 2. Rate the additional support necessary, in terms of <u>physical</u> resources, for a newly enrolled student with a severe/mild/no/ intellectual disability, who makes no/a modest amount/a great deal of/ effort at classwork.
- 3. Rate the additional support necessary, in terms of <u>personnel</u> resources, for a newly enrolled student with a severe/mild/no/intellectual disability, who makes no/a modest amount/a great deal/ of effort at classwork.

Instructions to teachers (Appendix C)

The following instructions were developed to be read out to the respondents:

"My name is Rory Mc Nally and I am currently undertaking a Masters degree in Special Education. Thank you very much for agreeing to take part in this short survey. As I have explained to your principal, I am attempting to measure the levels of additional classroom support that you think are necessary for specific students in your class. Your responses and the school's identity will remain anonymous and at no time will you have to declare any personal details. Your school will receive a copy of the overall study. You should have 4 sheets in front of you – a student profile, Measurement A, Measurement B and Measurement C.

At the end of each short task, please indicate that you have completed it by putting your pen down.

We'll begin with the profile of a hypothetical student, Brian, who could be a potential student in your class next term. <u>Please read carefully through the profile in front of you.</u>

Please turn to Measurement A - this measures additional physical support in respect of Brian.

By physical support I am talking about <u>additional</u> curriculum resources, textbooks, remediation and extension materials, enrichment programmes, computers and software etc.

If you circle a 7, you feel that Brian needs the maximum amount of <u>additional physical support</u>, if you circle a 4, you think he needs moderate <u>additional physical support</u> and if you circle a 1, you think he needs minimal <u>additional physical</u> support.

Please complete Measurement A only

Now we turn to Measurement B - this measures additional personnel support in respect of Brian..

By additional personnel support, I am talking about aides, paraprofessionals or volunteers. Circle the 7, if you feel that Brian requires an aide, paraprofessional or volunteer for six hours per day, circle a 4 for three hours per day or circle a 1 for approximately one hour per day. Please feel free to refer to your profile again and now complete Measurement B.

Now turn to Measurement C.

You will be asked to <u>draw lines</u>. Lines go from left to right, starting at the dot on the left hand side of the page. <u>Please do not go as far as the edge of the</u> <u>paper</u>. Lines of different length indicate different levels of additional support

Please look at item 1 which requires a linear response to the level of <u>additional</u> support you think a student of <u>average ability</u> would need. By <u>additional</u> support here, I mean a combination of <u>physical</u> and <u>personnel</u> support. For example, a line up to a centimeter long would indicate <u>minimal</u> additional support, a line towards the centre of the page indicates <u>moderate</u> additional support and a line towards the edge of the page would indicate <u>maximum</u> additional support. (Researcher models).

Please complete item 1.

Now, please indicate your level of perceived <u>additional</u> support for Brian, as described in items 2 & 3.

Thank you very much for your time."

Trial and amendments

Ethical permission was obtained from the University Ethics Committee to conduct the study. The seven-point scale and the magnitude scaling measurements were given to 17 professional adults in order to ascertain the reliability of the measurement tools. Some alterations were necessary, based on the results of this trial. Two items were deleted from the second section of the questionnaire (Measurement B), as there appeared to be some ambiguity and overlapping in the content of both items. Excluding these two items, Cronbach's Alpha indicated a reliability of 0.81 for Measurement A, a reliability of 0.76 for Measurement B and a reliability of 0.76 for both sections of the questionnaire combined.

The original vignette contained the term 'learning disability' and this caused some confusion to the respondents, some of whom wanted a less

ambiguous term. As a result, this term was replaced with the term 'intellectual disability'. The researcher found that during the trial it was necessary to establish when the respondents were finished each measurement tool. This resulted in the inclusion of a request for respondents to put their pens down when each section was completed. The researcher also underscored words and phrases in the 'instructions to teachers' section that required emphasis. The word 'vignette' was replaced by the term 'student profile' in the instruction section. This was to facilitate a quicker understanding of the purpose of the vignette on the part of the respondents.

Subjects in the study

The total Catholic Primary School teaching staff in Western Australia numbered 1628 in 1997. Three hundred and thirty of these were male and 1298 female, making the then ratio of male to female teachers almost 1:4. This study surveyed 72 female regular classroom teachers from nine schools. A cluster of nine Catholic Primary schools was chosen from the same metropolitan area as a convenience sample and all female classroom teachers were invited to participate in the study. The researcher did not include male teachers, as he did not want to complicate the study by leaving it open to a possible gender effect.

Seventy-two teachers read a vignette on the hypothetical student. They then responded to two sections of a questionnaire. Measurement A sought responses (using a seven-point Likert scale) to teachers' expected levels of additional physical classroom support for the hypothetical student. Measurement B sought responses to their perceived need for additional

personnel support for the student. A secondary tool, by means of magnitude scaling (Measurement C), was employed to measure regular teachers' attitudes to both physical and personnel support for integrated students with disabilities. Measurement C sought a quantitative measurement of the teachers' expected additional support levels (physical and personnel) for the hypothetical student.

Study Design

The design of this study included three independent variables (school, ability and effort). Eight schools participated in the study. They were all Catholic schools located in the eastern-metropolitan area of Perth. All of these schools have a similar fee structure. It can thus be assumed that each school had a similar cultural, historical and religious student make-up. Because of the similarity in the type of schools used in this study, it was not anticipated that a school factor would be a critical variable in this study. Because of the similarity of schools, it was anticipated that it would not matter which of the schools the teachers worked in analysing their responses to the measurement tools. If an interaction between the schools and other factors were to be revealed, a multi-level analysis would be performed on the data.

Ability was divided into three levels. One level was established as average intellectual ability. This refers to a student who has normal cognitive functioning and would be expected to perform academic tasks at a moderate level. A second level of ability was established as mild intellectual ability. This refers to mental development at between one-half and three-

quarters of the normal rate and which has been assessed at an IQ between 50 and 69. The third level was severe intellectual disability and this is defined as mental development at less than one-quarter of normal cognitive growth and assessed at an IQ of between 20 and 40.

The effort variable was also divided into three levels; low, moderate and high. <u>Effort</u> refers to the level of input that a student expends in the areas of classwork, assignments and homework. Ability and effort were matched in each level to form a nine-cell design (See Figure 3.1).





Figure 3.1

Study Design

Procedure

The principal of each of the nine schools in the study was contacted and permission sought to meet with nine female teachers from each school. They were told that the project would attempt to measure teachers' expectations with regard to additional physical and personnel classroom support for a hypothetical student. The researcher compiled a list of 72 eligible teachers from the nine schools and put them in alphabetical order by surname. A vignette was then randomly assigned to each name on the list. The number of teachers varied from school to school and because of the random selection method employed, it was possible that two or more teachers from the same school responded to measurements on the same vignette.

The researcher made arrangements with the principal of each school to meet the teachers in groups. He met with four groups prior to the commencement of school, three groups after school and two groups at the commencement of their scheduled staff meeting. Meeting locations varied. In some schools the researcher was required to use the staffroom, in other schools, the library and in one school, the classroom. The researcher had to revisit two of the schools, one because two staff members were unable to attend on the appointed day, the other because the results of two of the respondents were identified as outliers and the teachers had to be replaced.

In each group, the researcher and the teachers invariably had a brief conversation prior to the commencement of the task. Questions were asked about the purpose of the study, the anonymity of the respondents, the expected responses and the length of time the task would take. The

researcher gave each teacher in the group the assigned vignette and Measurements A, B and C, and read the instructions carefully and slowly for each measurement. The difference between additional physical resources and additional personnel resources was explained and any questions answered. Use of the Likert scale was explained, even though all respondents seemed to be familiar with it.

The correct response method to the magnitude-scaling tool only became clear when the researcher physically modeled how to respond to it. The researcher placed great emphasis on the fact that item1 in this measurement referred to the expected levels of additional support for an average student making a modest amount of effort and that items 2 and 3 referred to the expected additional support levels for Brian. This modeling was repeated to ensure that all respondents were clear as to how to respond to this measurement. The eight respondents, who were assigned a vignette describing a student of average ability making a modest amount of effort, were reminded that there was no difference between the student described in item 1 and the student in their vignette.

Response rate

The response rate was 100%. The procedure employed in this study guaranteed a full response rate. Because the researcher was physically present to collect the data, there was no possibility of the teachers forgetting to fill in the responses or having to post them. When teachers could not make the appointed time, the researcher simply returned to the school at a later date to collect the data. As all schools were located in the same

metropolitan area as the researcher lives in, the collection of data was made easier.

Hypotheses

HO¹: There will be no significant difference in regular classroom teachers' attributions of the need for additional support for students with and without intellectual learning disabilities.

H1¹: There will be a significant difference in regular classroom teachers' attributions of the need for additional support for students with and without intellectual learning disabilities.

HO²: There will be no significant difference in regular classroom teachers' attributions of the need for additional support for students exhibiting different levels of effort.

H1²: There will be a significant difference in regular teachers' attributions of the need for additional support for students exhibiting different levels of effort.

HO³: There will be no significant interaction between the factors of ability and effort with respect to needed support.

H1³: There will be significant interaction between the factors of ability and effort with respect to needed support.

The level of significance used for the various statistical tests was set at 0.05. The null hypothesis will be rejected if this level of significance is attained for the particular variable being tested. The experimental hypothesis will be rejected if the null hypothesis for the same variable is accepted.

Chapter 4

RESULTS

In this chapter, the results of the study are presented. The data were initially analysed using multivariate procedures (Wilks' Lambda). These revealed no interaction between the schools factor and the other independent variables, ability and effort. A multivariate analysis of variance was then employed to test for main effects of ability and effort on the dependent variables for the seven-point scale and the magnitude scaling. Both scales indicated a main effect for ability. Univariate analyses were used to test for differences between groups on both the seven-point and magnitude scaling instruments. A Scheffé test of multiple comparisons was used to explore differences between the three levels of ability for both scales. The data are presented on tables and figures for each of the measurement instruments. The results are then briefly summarised.

Outliers

It was necessary to examine the data for outliers. An outlier is a term used in statistical data and refers to extreme cases on one variable or a combination of variables that distort the pattern of data, with no logical reason being evident for these extremes. These data can be omitted from the study (Tabachnick & Fidell, 1996). Two participants were identified as possible outliers. In the seven-point instrument, one participant returned an

extraordinarily high level of expected additional support (both physical and personnel) for an average-ability student making a modest effort in class. This participant's expected additional support levels (average score = 5) were in contrast to the score of the other seven participants' responses for the same student (average score = 1.26). This participant returned a similar expected support level (average score = 1.09) as the other seven subjects (average score = 1.1) for the magnitude-scaling instrument for this same student.

Similarly, a second participant returned an unusually high level of expected additional support (physical and personnel) in the seven-point instrument for an average-ability student who was reported to make a low effort in class. This participant's expected additional support levels (average score = 5.8) are at odds with the seven other subjects' mean expected additional support levels (average score = 2.3) for the same student. This participant also returned a similar (average score = 1.84) expected additional support level as the other subjects (average score = 1.62) for this same student.

These data indicated that these two participants could have misunderstood the criteria and required procedure for completing the sevenpoint instrument. They indicated a very high level of additional support for an average student, which was at odds with their expectations of the same additional support for the same student in the magnitude-scaling instrument. Both participants were defined as outliers and their results were not included in the study. Data were then collected from two different participants under the same conditions as prescribed for those excluded from the analysis.

These two new participants were taken from the original list of randomly assigned teachers. The data were then analysed using multivariate procedures. The independent variables were ability, effort and school. The multivariate analysis yielded no significant interactions among ability, effort and schools.

These outliers were not replaced to strengthen the results of the study or to 'suit' the researcher's data. The researcher could find no logical explanation for these extremes of additional support for students of average ability and thus chose to exclude these data from the study.

Reliability and validity

Cronbach's Alpha indicated a reliability of 0.81 for the dependent variable physical support, 0.76 for personnel support and a combined reliability of 0.76 for both dependent variables. Content validity was determined as a result of the trial using 17 professionals. They indicated that the measures covered the definition of each variable. With regard to construct validity, appropriate alterations were made to the items and instructions and the questionnaire was based on the well-established Likert scale, which efficiently distinguished high and low levels of attitude of the respondents. The random assigning of vignettes to teachers, the fact that each teacher received similar instructions and measurement tools, the fact that each response was scored in the same fashion and each teacher had an equal opportunity to ask questions if they were unsure about any aspect of the study gives these results an implicit acceptable validity
The schools variable was not included in any further analyses as no major differences in ability and effort could be ascertained among the individual schools (See Table 4.1). The schools factor was not involved in any significant multivariate three-way interaction ($\mathbf{F} = 0.522$, $\mathbf{p} > 0.01$). The schools factor was not involved in any significant interaction with the ability factor ($\mathbf{F} = 0.617$, $\mathbf{p} > 0.01$), nor with the effort factor ($\mathbf{F} = .474$, $\mathbf{p} > 0.01$). The relatively uniform level of mean responses among schools was not a surprising result and this confirmed the researcher's earlier prediction that there would be no interaction between the schools factor and the other independent variables.

Table 4.1Multivariate tests of main effects andinteractions for schools/ability/effort (N = 72)

| Effect | Value | F | Error df | Sig |
|--|-------|-------|----------|-------|
| Ability x School | | | | |
| Wilks' Lambda | 0.610 | 0.617 | 44.000 | 0.878 |
| Effort x School Wilks' Lambda | 0.677 | 0.474 | 44.000 | 0.963 |
| Ability/ Effort x School Wilks' Lambda | 0.544 | 0.522 | 44.000 | 0.968 |

A multivariate analysis of variance was performed on the two dependent variables, additional physical support and additional personnel support, for the factors ability and effort. SPSS MANOVA was used for the analyses of main effects. The total N was 72 teachers. Wilk's criterion indicated no significant multivariate interactions between the factors ability and effort. There was no significant effect for effort ($\underline{F} = 1.349$, $\underline{p} > 0.01$), however, the multivariate results yielded one main effect (See Table 4.2). A significant effect ($\underline{F} = 19.866$, $\underline{p} < .001$, df = 71) was noted for ability. No significant effect was noted for the interaction of ability and effort ($\underline{F} =$ 0.781, $\underline{p} > 0.01$).

Table 4.2Multivariate tests of main effects and interactionsusing seven-point scale for ability/effort (N=72)

| Effect | Value | F | Error df | Sig |
|------------------|-------|--------|----------|-------|
| Ability | | | | |
| <u> </u> | | | | |
| Wilks' Lambda | 0.371 | 19.866 | 124.000 | 0.000 |
| | | | | |
| Effort | | | | |
| Wilks' Lambda | 0.918 | 1.349 | 124.000 | 0.256 |
| | | | | |
| Ability x Effort | | | | |
| Wilks' Lambda | 0.906 | 0.781 | 124.000 | 0.620 |

Seven-point scale

Regular classroom teachers perceive that students of average ability would require significantly less additional physical resources than students with a mild intellectual disability ($\underline{MD} = 1.7083$, p < .001, df = 71) or a severe intellectual disability ($\underline{MD} = 2.2500$, p < .001, df = 71). Table 4.3 displays the relevant data. Scheffé <u>post hoc</u> comparisons were used to compare means.

Figure 4.1A depicts the average score for those of average, mild and severe ability, in regard to perceived additional physical support levels for the seven-point data. There appears to be an overall difference in the pattern of data at the mild and severe levels of ability indicating a possible interaction, however this difference indicated a non-significant result (p > 0.01).





Figure 4.1B graphs the differences between all ability levels in regard to additional personnel support for the seven-point data. Univariate tests indicated that the differences between all ability levels were significant. Again, despite the appearance of an interaction in the pattern of data, the disparity was not sufficient to indicate a significant interaction (p > 0.01).







The data in table 4.4 show the seven-point instrument in regard to additional personnel support. There were significant differences between all the profiled levels of ability. Teachers perceived that students of average ability would require significantly less additional personnel support than students with mild ($\underline{MD} = 1.2423$, $\underline{p} < .004$, df = 71) and severe intellectual disabilities ($\underline{MD} = 2.2111$, $\underline{p} < .001$, df = 71). Students with mild intellectual disabilities were perceived to require significantly less additional personnel support than students with severe intellectual disabilities ($\underline{MD} = 0.9688, p < .024, df = 71$).

Table 4.4 Comparison of ability levels for the seven-point data

| Dependent | (I) | (J) | Mean | Std. | |
|------------|---------|---------|------------------|-------|-------|
| variable | Ability | Ability | ility Difference | | Sig. |
| <u> </u> | Average | Mild | -1.7083* | 0.238 | .000 |
| Need for | | Severe | -2.2500* | .238 | .000 |
| additional | | | | | |
| physical | Mild | Average | 1.7083* | 0.238 | 0.000 |
| resources | | Severe | -0.5417 | 0.236 | 0.094 |
| | Severe | Average | 2.2500* | 0.238 | 0.000 |
| | | Mild | 0.5417 | 0.236 | 0.094 |
| Need for | Average | Mild | - 1.2423* | 0.329 | 0.004 |
| additional | | Severe | - 2.2111* | 0.329 | 0.000 |
| personnel | Mild | Average | 1.2423* | 0.329 | 0.004 |
| resources | | Severe | - 0.9688* | 0.325 | 0.024 |
| | Severe | Average | 2.2111* | 0.329 | 0.000 |
| | | Mild | 0.9688* | 0.325 | 0.024 |
| | | | | | |

*. The mean difference is significant at the 0.05 level.

In summary, the findings from the seven-point data indicate that regular classroom teachers perceive a significant difference in the levels of additional physical support only between students with average ability and mild/severe intellectual disabilities. They do not perceive the need for a difference in the levels of additional physical support between students with mild and severe learning disabilities. However, when it comes to additional personnel support for the same students, regular teachers see the need for progressively more additional support as the level of disability of the target student increases.

Magnitude Scaling

The magnitude scaling data were analysed next (See Table 4.3). There was no evidence of any significant effect for the interaction of ability and effort ($\underline{F} = .535$, $\underline{p} > 0.01$). The combined dependent variables (personnel and physical support) were significantly affected by the ability factor ($\underline{F} = 7.062$, $\underline{p} < .000$, df = 71). The effort variable was not significant ($\underline{F} = 1.061$, $\underline{p} > 0.01$).

The multivariate tests confirmed the importance of the ability dimension. Univariate analyses were used to explore the major differences between groups on both the seven-point scale and the magnitude-scaling instrument.

Table 4.3 Multivariate tests of main effects and interactions

Value F Error df Sig Effect Ability 0.000 0.371 7.062 44.000 Wilks' Lambda Effort Wilks' Lambda 0.832 1.061 46.000 0.387 Ability x Effort Wilks' Lambda 0.830 0.535 44.000 0.823

using magnitude scaling for ability/effort (N=72)

Scheffé's contrasts were applied to the magnitude scaling instrument data (See Table 4.5). Again regular classroom teachers perceived that students of average ability would require significantly less additional physical resources than students with a mild intellectual disability ($\underline{MD} = -3.4363$, $\underline{p} < .013$, df = 71) or a severe intellectual disability ($\underline{MD} = 5.9433$, $\underline{p} < .001$, df = 71). Tests on the differences between levels of ability also revealed that regular classroom teachers perceived little difference ($\underline{MD} = 2.5070$, $\underline{p} > 0.01$) in regard to the need for additional physical support for students with mild and severe intellectual disabilities. These data are similar to the findings from Table 4.4.

Students with average ability were perceived to require significantly less additional personnel support than students with severe (MD = 7.0932, p < .001, df = 71) intellectual disabilities. Students with mild learning disabilities were seen to require significantly less additional personnel support than students with severe intellectual disabilities (MD = 4.0936, p < .003, df = 71). These results indicate that regular classroom teachers perceived a significant difference between mild and severe levels of ability when they made judgements between the levels of additional personnel support for students in their class. The data for the magnitude-scaling instrument confirm the findings of the seven-point data in regard to additional personnel support.

TABLE 4.5 Comparisons of ability levels for magnitude scaling

| Dependent | (I) | (J) | Mean | Std. | Sig. |
|------------|---------|---------|------------|-------|-------|
| variable | Ability | Ability | Difference | Error | |
| Need for | Average | Mild | -3.4363* | 1.059 | 0.013 |
| additional | | Severe | -5.9433* | 1.059 | .000 |
| physical | | | | | |
| resources | Mild | Average | 3.4363* | 1.059 | .013 |
| | | Severe | - 2.5070 | 1.059 | 0.081 |
| | Severe | Average | 5.9433* | 1.509 | 0.000 |
| | | Mild | 2.5070 | 1.059 | 0.081 |
| Need for | Average | Mild | - 2.9995* | 1.057 | 0.032 |
| additional | | Severe | - 7.0932* | 1.057 | 0.000 |
| personnel | | | | | |
| resources | Mild | Average | 2.9995* | 1.057 | 0.032 |
| | | Severe | - 4.0936* | 1.057 | 0.003 |
| | Severe | Average | 7.0932* | 1.057 | 0.000 |
| | | Mild | 4.0936* | 1.057 | 0.003 |
| | | | | | |

*. The mean difference is significant at the 0.05 level.

These findings support the findings from the analysis of the sevenpoint data. The magnitude scaling data revealed that regular classroom teachers perceived the need for a significant difference between students of average learning ability and students with mild or severe intellectual disabilities, in regard to additional physical support. They did not perceive a difference in the levels of additional physical support required for students with mild and severe intellectual disabilities. However, when regular classroom teachers made judgements about the levels of personnel support for students in their class, they perceived a significant difference between each of the three identified levels.

Figure 4.2A shows the differences for ability between average and mild and average and severe, in regard to the perceived need for additional physical for support for the magnitude-scaling instrument. There appears to be a considerable disparity between patterns of the levels of effort reported for the levels of ability, but no significant interaction was indicated (p > 0.01).

Figure 4.2B confirms the trend depicted in the previous graph by showing the differences between all ability levels in regard to additional personnel support. The data indicate a high concordance with the overall pattern of results reported in Figure 4.1B.





Score on physical items for ability/effort levels





for magnitude scaling

Summary of results

Results of this study suggest that regular classroom teachers see ability as a significantly important dimension when assessing the additional support levels they perceive necessary to for particular students in their classes. They perceive students with intellectual disabilities as needing more additional support than students of average ability, regardless of the effort expended by the student. However, with regard to additional physical support, they do not see a difference in the levels of support needed for students with mild and severe intellectual disabilities. With regard to additional personnel support, regular classroom teachers perceive a significant difference between all levels. The results of both the seven-point data and the magnitude scaling data strongly support these statements. These results are similar to the findings of Clark (1997), Weiner(1979) and Weiner and Kula (1970), who all found that ability was the major determining factor when teachers make judgements about students. The results of this study are at odds with Graham (1990) who found that effort and ability were linked to teachers' perceptions of support levels for students with intellectual disabilities.

HO¹ was rejected as a significant difference was noted in regular classroom teachers' attributions of the need for additional support for students with and without intellectual disabilities. HO² was accepted as no significant difference was noted in regular classroom teachers' attributions of the need for additional support for students exhibiting different levels of effort. HO³ was rejected as a significant interaction was noted between the factors of ability and effort with respect to needed support.

Chapter 5

DISCUSSION

Summary

This study set out to examine the attitudes of regular classroom teachers to perceived additional support levels for integrated students with intellectual disabilities. The dependent variables were additional physical and additional personnel support. The independent variables were schools, effort and ability. The design of the study matched the three levels of ability with the three levels of effort, creating a nine-cell design. The schools factor was not subjected to further analyses once it had been established that there were no significant differences between schools and that the schools factor did not interact with either of the other two independent variables. The subjects comprised 72 regular classroom teachers, employed in Catholic schools in an eastern metropolitan area of Perth, Western Australia. The measurement tools used were a seven-point scaled questionnaire and a magnitude-scaling instrument. Reliability and validity were established for these measurement tools.

The results of the seven-point data and the magnitude scaling were highly consistent with each other for both dependent variables. This study found that a student's ability is a determining factor on a regular classroom teacher's attributions of the student's performance. The results also support the findings of leading advocates of attribution theory. The results of this study concur with the findings of Clarke (1997), Weiner (1979) and Weiner and Kula (1970), who all reported that when teachers make judgements on student performance and support levels needed, they consider the ability of

the student a major influencing factor. Results of this study also indicated that regular classroom teachers do not consider the effort that a student expends on classwork as a major factor when making judgements about the levels of support that student needs. These findings are at odds with Graham (1990) who reported that effort was strongly linked to teacher's perceptions of support levels for students with intellectual disability. Regular classroom teachers see students with intellectual disabilities as requiring more additional physical support than their peers of average ability but they do not see any difference in the levels of additional support needed for students with mild and severe intellectual disabilities. However, when it came to additional personnel support, regular classroom teachers perceived a significant difference between the three levels of ability.

Conclusions

- There were no significant differences in ability and effort between the schools used in this study
- Regular classroom teachers do not take student effort into account when making judgements on the levels of support needed for that student.
- Regular classroom teachers do not see a difference in the levels of additional physical support materials needed for students with mild and severe intellectual disabilities.
- When ascertaining the need for additional personnel support, regular classroom teachers see a significant difference in the levels of support needed for average ability students, students with mild

intellectual disabilities and students with severe intellectual disabilities.

• Regular classroom teachers rate additional personnel support more beneficial than additional physical support for students with severe disabilities.

Limitations of study

This study has been limited to a small cluster of Catholic primary schools in a few neighbouring suburbs of Perth, Western Australia. Would the results be different in Catholic schools in different suburbs? Does the Catholic ethos influence the judgements of Catholic teachers of students who, because of an intellectual disability, require additional, specialised academic supports? Would a similar study in Anglican or State primary schools yield the same results? While the researcher acknowledges that the results cannot be generalised to all Catholic primary schools in Perth, in Western Australia or in Australia, there is little reason to suspect that Catholic teachers in other parts of Australia would cause these results to differ greatly. Similarly, the researcher acknowledges that the results of this study cannot be generalised to other privately run primary schools and public primary schools in Perth, in Western Australia or in Australia. Again the researcher has no reason to consider that there might be any significantly different results in a similar study in these schools.

Another limitation of this study is that all of the subjects were female and therefore, technically, the results are not representative of the general population of primary teachers in Perth, in Western Australia or in

Australia. However as female teachers significantly outnumber male teachers in private and public primary schools, it was deemed appropriate to use females only as the subjects in this study.

The researcher acknowledges that the measurement tools used in the study were less than sophisticated. There were only four items in one of the questionnaires. More items might have yielded an even higher reliability. However, the researcher is satisfied that the number of items measured what needed to be measured, was user friendly to the very busy teachers involved in the study and was appropriate to the level of this study.

This study was limited to measuring only two of the many additional supports available to regular classroom teachers for the successful inclusion of students with intellectual disabilities. The researcher acknowledges that regular classroom teachers may also receive additional supports in the form of class size reduction, extra administration time and professional development.

Implications for administrators

This study has implications for educational administrators here in Western Australia, perhaps especially those employed by the Catholic Education Office, in that it provides basic guidelines for the allocation of funding for students with intellectual disabilities. Catholic schools are currently funded for special education courtesy in part of Commonwealth and State grants, the shortfall made up out of the school budget. The study results indicate that regular classroom teachers see additional personnel support as being very important for the successful inclusion of students with

severe intellectual disabilities. They are the personnel mainly responsible for the success of the inclusion of students with intellectual disabilities. Their concerns should therefore be, at the very least, considered.

The study has implications for Commonwealth and State funding programmes for special education. The results suggest that the focus of funding for special education should be diverted to providing trained personnel to support regular classroom teachers in their bid to make inclusion successful. This, in turn, has implications for the directing of appropriate funding to adequately train personnel in the area of special education, rather than spend money on computers, software and remediation kits for students with severe intellectual disabilities. The results of this study suggest that students with intellectual disabilities are going to be more successfully included in the regular classroom as a result of having access to trained supportive personnel.

Educational administrators could use the results of this study to help them determine the type and level of additional support afforded to students with intellectual disabilities and to heighten awareness of the needs of students with intellectual disabilities. In the light of this study, Catholic Education Office administrators in Western Australia should perhaps reassess the method in which funds for special education are distributed and take into account the attributions of their classroom teachers with regard to additional support levels for integrated students with intellectual disabilities. The administrators might consider providing appropriate training for regular classroom teachers, in light of the section of the results that indicates regular

severe intellectual disability are no different to those for students with a mild intellectual disability.

Implications for teachers

A surprising result of this study was the fact that regular classroom teachers perceived no difference in the additional levels of physical support needed for students with mild and severe intellectual disabilities. This would seem to suggest that regular classroom teachers would use the same level and type of physical resources to support students with severe intellectual disabilities as students with mild intellectual disabilities. One possible explanation for this result is that regular classroom teachers grab the opportunity of receiving any additional physical resources to support students with intellectual disabilities and what matters primarily is to get the resources. Who it's for can be sorted out later.

Another more serious scenario is that regular classroom teachers are unsure about the different resource needs of students with mild and severe intellectual disabilities. This has implications for a heightening of awareness among regular classroom teachers of the needs of students with mild, moderate and severe intellectual disabilities. Regular classroom teachers would improve the level of success for the inclusion of students with intellectual disabilities if they could access appropriate seminars which delineated the types and levels of additional support most suited to the various levels of students with intellectual disabilities.

The results of this study also give guidelines to the regular classroom teacher in how best to use additional support that is available. Students with

severe intellectual disabilities will benefit more from additional personnel support as opposed to additional physical support. These results, coupled with their attendance at the seminars alluded to in the previous paragraph will ensure that the allocation of additional support for students with intellectual disabilities is optimally utilised.

Implications for schools

The implications of the results of this study for schools in Western Australia is connected to the type and level of additional support that is allocated to target students in the school, assessing the benefits of such support and redirecting additional resources to employing trained personnel to support students with intellectual disabilities. Perhaps the most salient implication for schools is that the concerns of regular classroom teachers in regard to additional support for integrated students with intellectual disabilities should be assessed and addressed so that students with intellectual disabilities can be successfully included in the regular class. If each Catholic school conducted a smaller, similar survey of its regular classroom teachers, then more informed decisions could be made about levels, types and recipients of additional support in the classroom.

Implications for research

The dearth of this type of study Australia-wide makes this study valuable as a starting point to examine more thoroughly regular teachers' attitudes to support levels for integrated students with intellectual disabilities or indeed to provide current data on the attitudes of the regular

classroom teacher to the whole notion of inclusion. With a growing number of advocates for fully inclusive schools in the U.S., it would seem beneficial to assess the current status of regular classroom teachers' attitudes to this concept here in Australia. Regular classroom teachers are the personnel who are most responsible for the occurrence of successful inclusion in schools. It would also be beneficial to explore the reasons why classroom teachers see a difference between the additional support needs of students with mild and severe only in respect to personnel support. This study could also be repeated in Western Australia's public primary schools and even extended to the secondary schools to compare results

REFERENCES

- Bacher, J.H., (1964). The effect of special class placement on the self- concept, social adjustment, and reading growth of slow learners. *Dissertation Abstracts*, 25, 7071 (New York University).
- Bar-Tal, D., (1979). Interactions of teachers and pupils. In I. H. Frieze, D. Bar-Tal and J. S. Carroll (Eds), New approaches to social problems, pp.337-358, San Francisco: Jossey-Bass.
- Braun, C. (1976). Teachers' expectations: sociopsychological dynamics. *Review of Educational Research*, 46, 185-214.
- Brown, L., Schwarz, P., Udvari-Solner, A., Kampschooer, E., Jolenson, F., Jorgenson, J. and Greenwald, L. (1991). How much time should students
 with severe intellectual disabilities spend in the regular education classrooms and elsewhere? *Journal for the Association of the Severely Handicapped*, 16(1), 39-47.
- Center, Y., & Ward, J. (1987). Teachers' attitudes towards the integration of disabled children in regular schools. *The Exceptional Child*, 31(1), 41-56.
- Childs, R.E. (1981). Perceptions of mainstreaming by regular classroom teachers who teach mainstreamed educable mentally retarded students in the public schools. *Education and Training of the Mentally Retarded*, 16, 225-227.
- Clarke, M.D., (1997). Teacher response to learning disability: A test of attributional principals. *Journal of Learning Disabilities*, 30, pp 69-79.
- Coates, R.D., (1989). The Regular Education Initiative and Opinions of Regular Classroom Teachers. *Journal of Learning Disabilities*, 22. (9), 532-536.
- Collins, M.K., (Chairman), (1984). Integration in Victorian Education. Melbourne: Department of Education.
- Cooper, H. M. (1979). Pygmalion grows up: a model for teacher expectation and performance influence. *Review of Educational Research*, 49, 3 89-410.
- Curriculum Council of Western Australia. (1998). *Curriculum Framework*. Osborne Pk., W.A.: Curriculum Council,

- de Lemos, M. M. (1994). Schooling for students with disabilities. Canberra: Australian Government Publishing Services.
- Department of Employment, Education and Training. (1990). A Fair Chance for All. Australian Government Publishing Service.
- Diggs, E.A., (1964). A study on change in the social adjustment among the blind. In P. Zahl (Ed.), *Blindness*, Englewood Cliffs, N.J.:Prentice-Hall.
- Dusek, J. B. (1975). Do teachers bias children's learning? *Review of Educational Research*, 45, 661-684. Perth: Department of Education.
- Education Department of Western Australia, (1993). Policy and guidelines for the education of students with disabilities. In *Social Justice in Education*. Perth: Department of Education.
- Education Department of Western Australia, (1994). Report of the Ministerial Task Force on the Education of Students with Disabilities and Specific Learning Difficulties. Perth: Department of Education.
- Evans, I. M. (1995). *Testing and diagnosis: A review and evaluation*. Baltimore: Paul H. Brookes.
- Fuchs, D., & Fuchs, L. (1994). Inclusive schools movement and the radicalisation of special education reform. *Exceptional Children*, 60(4), 294-309.
- Fulk, B.J.S., & Hirth, M.A. (1994). Perceptions of special education program effectiveness and attitudes towards inclusion. Paper presented at the American Educational Research Association Conference, New Orleans, April, 1994.
- Gallagher, P.A. (1985). Inservice: A mandated special education course and its effects on regular classroom teachers. *Teachers Education and Special Education*, 8, 188-197.
- Gans, K.D., (1987). Willingness of Regular and Special Educators to Teach Students with Handicaps. *Exceptional Children, Vol.54, No.1, 41-*45.
- Garver-Pinhas, A., & Schmelkin., L.P. (1989). Administrators and teachers' attitudes towards mainstreaming. *Remedial and Special Education*, 10(4), 38-43.

- Graham, S. (1990) On communicating low ability in the classroom: Bad things good teachers sometimes do. In S. Graham and V. Folked (Eds.), Attribution theory: Applications to achievement, mental health and interpersonal conflict (pp.17-36). Hillsdale, NJ: Erlbaum.
- Graham, S., & Weiner, B. (1986). From attribution theory to developmental psychology: A round trip ticket? *Social Cognition, 4,* 152-179.
- Hallahan, D. P, & Kauffman, J.M. (1980). Exceptional Children: Introduction to Special Education. London: Prentice-Hall International Ltd.
- Harasymiw, S. P., & Horne, M. D., (1976). Teacher attitudes to integration. Journal of Special Education, 10, 393-401.
- Harvey, D.H.P. (1992). Integration in Victoria: Teachers' attitudes after six years of a no-choice policy. *International Journal of Disability, Development and Education*, 39, 33-45.
- Heider, F. (1958). The psychology of interpersonal relationships. New York: Wiley.
- Horne, M. D., 1985. Attitudes toward handicapped students: Professional, Peer and Parent Reactions. Lawrence Erlbaum Assoc. Hillsdale: New Jersey.
- Hudson, F., Graham, S. Warner, M. (1979). Mainstreaming: An examination of the needs of regular classroom teachers. *Learning Disability Quarterly*, 2, 58-62.
- Johnson, G.O. (1962). Special education for the mentally handicapped A paradox. *Exceptional Children*, 29, 62-69.
- Kern, W.H. & Pfaffle, H (1962). A comparison of social adjustment of mentally retarded children in various education settings. American Journal of Mental Deficiency, 67, 407-413.
- Kirk, S., Gallagher J. and Anastasiow, N. (1997). Educating Exceptional Children. (8th Edition). Boston, New York: Houghton Mifflin Company.
- Larrivee, B, & Cook L. (1979). Mainstreaming a study of the variables affecting teacher attitude. *The Journal of Special Education*, 13(3), 315-324.

- Lipsky, D.K., & Gartner, A.(1991). Restructuring for quality. In Loyd, I., Repp, D., & Singh, K. (Eds.), *The Regular Education Initiative: Alternative perspectives, issues and models* (pp43-56). Sycamore, IL: Sycamore.
- Lodge, M, (1981). Magnitude Scaling: Quantitative Measurement of Opinions. California, USA: Sage Publications Inc.
- Myles, B.S., & Simpson, R.L. (1989). Regular educators modification preferences for mainstreaming mildly handicapped children. *The Journal of Special Education*, 22(4), 479-491.
- Parenter, T.P. & Nash, R. (1987). Attitudes of teachers and parents in the A.C.T. towards the integration of moderately intellectually handicapped children. *Australasian Journal of Education*, 11(2), 26-31.
- Schultz, L. R. (1982). Educating the special needs student in the regular classroom. *Exceptional Children*, 48, 366-368.
- Scruggs, T., & Mastropieri, M. (1996). Teacher perceptions of mainstreaming/inclusion, 1958-1995: A research synthesis. *Exceptional Children, Vol. 63,No. 1, pp. 59-74.*
- Sigafoos, J. & Elkins, J. (1994). Concerns of teachers about the integration of children with physical versus multiple disabilities. *Australiasian Journal of Special Education*, 18(2), 50-56.
- Stainback, S., & Stainback, W.(1992). Curriculum considerations in inclusive classrooms: Facilitating learning for all students. Baltimore: Paul Brookes.
- Tabachnick, B. & Fidell, L. (1996). Using Multivariate Statistics. New York :Harper Collins College Publishers.
- Teacher Opinion Poll (1975) Professional satisfaction. Today's Education, 64, 14.
- Vaughn, S., Schumm, J.S., Jallad, B., Slucher, J., & Saumell, L. (1996). Teacher's views of inclusion. Learning Disabilities Research and Practice, 11, 96-106.
- Wang, M.C, Reynolds, M.C. & Walberg, P. (1986). Rethinking special education. *Educational Leadership*, 44(1) 26-31.

- Ward. J., Bochner, S., Center, Y. Outhred, L., & Pieterse, M. (1987). Educating children with special needs in regular classrooms: An Australian perspective. Special Education Centre, Macquarie University, NSW.
- Warnock, H.M. (1978). Special Education needs Report of the Committee of enquiry with the education of handicapped children and young people. London: Her Majesty's Stationery Office.
- Weiner, B. (1979). A theory of motivation for some classroom experiences. Journal of Educational Psychology. 71, 3-25.
- Weiner & Kula. (1970). An attributional analysis of achievement motivation. *Journal of Personality and Social Psychology*, 15, 1-20.
- Weiner, B. (1986). An attributional theory of motivation and emotion. New York: Springer-Verlag.
- Western Australia. (1993). Report of the Ministerial Task Force on the Education of Students with Disabilities and Specific Learning Difficulties. Perth: Ministry of Education.

World wide web address sites for all information pertaining to inclusion on European and Asian countries in the 'Global perceptions' section of Chapter Two.

Centre for Studies on Inclusive Education. (1994). Provision for children with special needs in the Asia region. World Bank Technical paper. http://inclusion.uwe.ac.uk/csie/senasia.htm

Comprehensive Law. (1986). Portugal. http://inclusion.uwe.ac.uk/csie/slmac.htm

Decree 319. (1991). Portugal. http://inclusion.uwe.ac.uk/csie/slmac.htm

Five Year Plan. (1990). India. http://inclusion.uwe.ac.uk/csie/senasia.htm

Policies and Guidelines for Special Education. (1994). Philippines. Section 5, Article 1. <u>http://inclusion.uwe.ac.uk/csie/senasia.htm</u>

Report of Special Education Review. (1993). Ireland. http://inclusion.uwe.ac.uk/csie/irl.htm

The Education Bill. (1997). Ireland. Special Needs Education. <u>http://inclusion.uwe.ac.uk/csie/irl.htm</u>

The Portaria, 1102/97. (1997). Portugal. Section 105. http://inclusion.uwe.ac.uk/csie/slmac.htm

U.K. Education Act. (1996). http://inclusion.uwe.ac.uk/csie/ukedlaw.htm

U.K. Education Reform Act. (1994). http://inclusion.uwe.ac.uk/csie/ukedlaw.htm

UNESCO (1994). The Salamanca Statement and Framework for Action on children with special educational needs in the Asia region. http:// www.unesco.org/education/educprog/index.html

UNICEF. (1989). The UN Convention on the Rights of the Child. London. http://inclusion.uwe.ac.uk/csie/senasia.htm

Appendices

A

STUDY EXPLANATION

Dear colleagues

For my Masters in Special Education, I am researching regular classroom teachers' attitudes to additional support levels for students with disabilities, who are mainstreamed. I am seeking your participation in this study as I feel that your expected levels of additional support for these students are crucial to their success in the classroom.

For this study, I require up to 9 female teachers who have had experience in the regular classroom. Your participation will take a maximum of 20 minutes, during which time, in a face to face session, I will explain the essence of the three measurement tools (short surveys) to which you will be asked to respond.

Your responses are totally anonymous. I am not interested in comparing the attitudes of different schools, nor am I seeking to make judgements on your personal knowledge in the area of special education. The aggregated results of the study will be made available to your school. If you have any queries please contact me on:

Phone/Fax -

I would gratefully appreciate your cooperation in this study. Please indicate your intent to your principal and we can make a suitable time to meet. Yours faithfully Rory Mc Nally

CONSENT FORM

I agree to participate in Mr Mc Nally's research surveys, as detailed in his covering letter. I reserve the right to withdraw from the study at any time.

SIGNED_____ DATE_____

INSTRUCTIONS TO TEACHERS

"My name is Rory Mc Nally and I am currently undertaking a Masters degree in Special Education. Thank you very much for agreeing to take part in this short survey. As I have explained to your principal, I am attempting to measure the levels of additional classroom support that you think are necessary for specific students in your class. Your responses and the school's identity will remain anonymous and at no time will you have to declare any personal details. Your school will receive a copy of the overall study.

You should have 4 sheets in front of you – a student profile, Measurement A, Measurement B and Measurement C. At the end of each short task, please indicate that you have completed it by putting your pen down. We'll begin with the profile of a hypothetical student, Brian, who could be a potential student in your class next term. Please read carefully through the profile in front of you.

Please turn to Measurement A - this measures additional physical support in respect of Brian. By physical support I mean <u>additional</u> curriculum resources, for example, textbooks, remediation and extension materials, enrichment programmes, computers and software etc. If you circle a 7, you feel that Brian needs the **maximum** amount of <u>additional</u> <u>physical</u> support, if you circle a 4, you think he needs **moderate** <u>additional</u> physical support and if you circle a 1, you think he needs **minimal** <u>additional physical</u> support. Please complete Measurement A only

С

Now we turn to Measurement B - this measures additional personnel support in respect of Brian. By additional personnel support, I am talking about aides, paraprofessionals or volunteers. Circle the 7, if you feel that Brian requires an aide, paraprofessional or volunteer for **six hours per day**, circle a 4 for **three hours per day** or circle a 1 for approximately **one hour per day.** Please feel free to refer to your profile again and now complete Measurement B.

Now turn to Measurement C. You will be asked to <u>draw lines</u>. Lines go from left to right, starting at the dot on the left hand side of the page. <u>Please do not go as far as the edge of the paper</u>. Lines of different length indicate different levels of additional support Please look at item 1 which requires a linear response to the level of <u>additional</u> support you think a student of **average ability** would need. By <u>additional</u> support here, I mean a combination of <u>physical</u> and <u>personnel</u> support. For example, a line up to a centimeter long would indicate <u>minimal</u> additional support, a line towards the centre of the page indicates <u>moderate</u> additional support and a line towards the edge of the page would indicate <u>maximum</u> additional support.

(researcher models)

Please complete item 1.

Now, please indicate your level of perceived <u>additional</u> support for Brian, as described in items 2 & 3.

Thank you very much for your time."

VIGNETTE

D

Dear colleague

Brian is a potential student for your class next term. Recent psychological testing indicates Brian has no / a mild / a severe intellectual disability compared to students of his age. He currently undertakes instruction in the core subject areas of Mathematics, English, Social Studies and Science. Brian participates in social activities and is aware of school rules.

He always / sometimes / rarely works hard in class, making a good deal of / a modest amount of / little effort to complete assignments and homework. He can participate in group work, likes soccer and has two pet rabbits of which he is very fond. Brian's parents are anxious that he adjusts well to his new school and hope that he can settle smoothly into his new environment.

Measurement A – physical support: By physical support I mean additional curriculum resources, for example, textbooks, remediation and extension materials, enrichment programmes, computers and software etc. to support specific students in your class.

NOTES: Measurement B – personnel support: By personnel support I mean <u>additional</u> personnel to support specific students in your class, for example, aides, paraprofessionals and volunteers.

In each questionnaire, you are asked to circle the numbered response that best reflects the level of <u>additional</u> support that you would expect in order to help Brian succeed in your class.

Seven-point measurement on additional physical support

E

(Measurement A)

1.How many <u>additional</u> resources will you need to give adequate support to Brian in **Mathematics** lessons?

1 2 3 4 5 6 7 Very few very many

2. How many additional resources will you need to give adequate support to

Brian in Reading lessons?

| 1 | 2 | 3 | 4 | 5 | 6 | 7 |
|----------|---|---|---|---|---|-----------|
| Very few | | | | | | very many |

3. How many <u>additional</u> resources will you need to give adequate support to

Brian in Science lessons?

| 1 | 2 | 3 | 4 | 5 | 6 | 7 |
|----------|---|---|---|---|------|------|
| Very few | | | | | very | many |

4. How many <u>additional</u> resources will you need to give adequate support to Brian in Art?

| | 1 | - | 5 | 5 | U | , |
|------|-------|---|---|---|---|-----------|
| Very | / few | | | | | very many |

5. How many <u>additional</u> resources will you need to give adequate support to Brian in **Religion** lessons?

1 2 3 4 5 6 7 Very few very many

6. How many additional resources will you need to give adequate support to

Brian in **Health** lessons?

 1
 2
 3
 4
 5
 6
 7

 Very few
 very many

Seven-point measurement on additional personnel support

F

(Measurement B)

1. How much additional personnel support would you need to help Brian in outdoor activities? 2 3 4 5 6 1 7 Very little very much 2. How much additional personnel support would you need to help Brian in Church activities? 1 2 3 4 5 6 7 Very little very much 3. How much additional personnel support would you need to help Brian in small-group activities? 2 3 4 5 6 7 1 Very little very much 4. How much additional personnel support would you need to help Brian in

school excursions?

| 1 | 2 | 3 | 4 | 5 | 6 | 7 | |
|-------------|---|---|---|---|---|---|-----------|
| Very little | | | | | | | very much |

Magnitude scaling measurement on additional physical and personnel support

(Measurement C)

- Rate the additional support necessary for a newly enrolled student with average learning ability who puts in an average amount of effort.
 - ٠
- 2. Rate the additional support necessary, in terms of <u>physical</u> resources, for a newly enrolled student with a severe/mild/no learning difficulty, who makes no/a modest amount/a great deal of effort at classwork.
 - •
- 3. Rate the additional support necessary, in terms of <u>personnel</u> resources, for a newly enrolled student with a severe/mild/no learning difficulty, who makes no/a modest amount/a great deal of effort at classwork.

Fac of Community Services, Education

